UNIVERSITY OF KWAZULU-NATAL

AN EXAMINATION OF THE BARRIERS IN THE EFFECTIVE USE OF OUTSOURCING IN THE PUBLIC SECTOR

By Mondli Msiya 215080154

A dissertation submitted in fulfilment of the requirements for the degree of Master of Commerce (Management)

> School of Management, IT and Governance College of Law and Management Studies

Supervisor: Dr. Obianuju E. Okeke-Uzodike 2020

Declaration

I, Mondli	Msiya,	declare	that,
-----------	--------	---------	-------

- (i) The research reported in this dissertation, except where otherwise indicated, is my original research.
- (ii) This dissertation has not been submitted for any degree or examination at any other university.
- (iii) This dissertation does not contain other persons' data, pictures, graphs or other information, unless specifically acknowledged as being sourced from other persons.
- (iv) This dissertation does not contain other persons' writing, unless specifically acknowledged as being sourced from other researchers. Where other written sources have been quoted, then:
 - a) Their words have been re-written but the general information attributed to them has been referenced and
 - b) Where their exact words have been used, their writing has been placed inside quotation marks and referenced.
- (v) Where I have reproduced a publication of which I am author, I have indicated in detail which part of the publication was actually written by myself alone and have fully referenced such publications.
- (vi) This dissertation does not contain text, graphics or tables copied and pasted from the Internet, unless specifically acknowledged and the source being detailed in the dissertation and in the references section.

Signed:

Dedication

I dedicate this work to my siblings and children. To my siblings, Nonjabulo, Sneziwe, Phindile and Sanele, I wish to say that your understanding, tolerance and encouragement during the long hours of my absence from you as I pursued my studies have been a valuable contribution towards the completion of this work. To my children Iminathi, and Melokuhle, I can only say that despite my awareness of your need for my adequate company with you, I still proceeded with this project, consequently depriving you of your inborn right to have the time you needed to be with me. I shall ever be indebted to you.

Acknowledgements

I wish to thank our Lord and Saviour for guidance. None of this would have happened without him. My appreciation also goes to my late grandmother, Mrs Bertha Ndlovu, for instilling the importance of education in me. In addition, I would like to thank my late sister, Miss Nokubonga Msiya, for her support during this study. The same goes to my friends, Mr Phathokuhle Mtambo and Mr Nkosinathi Mhlongo, for their support.

I would like to express my gratitude to my supervisor, Dr. Obianuju E. Okeke-Uzodike. Thank you for your support and guidance throughout this dissertation. Furthermore, I would like to thank Miss Irene Loubser, who has been more than an assistant in my school work. Lastly, my greatest gratitude goes to the South African Social Security Agency for welcoming me warmly and allowing me to interview their employees.

Abstract

The government of South Africa has strived to centralize the procurement of goods and services, to address inefficiencies in the outsourcing of non-core activities in the public sector. This is deemed necessary to improve the efficiency of outsourcing processes in the public sector. Public sector outsourcing has faced various challenges, including unethical practices such as corruption, fraud and price inflation, resulting in inefficiency and increased operational costs. This study examined the barriers facing the outsourcing of functions at the South African Social Security Agency, Umzumbe Local Office. The design of this study was of an exploratory nature and the researcher used a qualitative approach and a non-probability, purposive sampling method. The sample size for the study comprised forty-one employees from the South African Social Service Agency, in Umzumbe Local Office. Primary data was collected via in-depth face-to-face interviews with six senior government officials, and focus group discussions involved thirty-five discussants. The research instrument used for the study was an interview schedule. Data was analysed using content analysis. The findings of the study reveal inefficiency in daily operations, a lack of accountability, poor communication and proper planning, non-compliance with supply chain laws, poor monitoring and evaluation mechanisms, and a shortage of capable human resources. As such, the outsourcing of activities in the public sector does not achieve efficiency in its operations. The study recommends the hiring of skilled personnel in the SCM unit, that the department must work closely with law enforcement agencies, and that a consequence management approach be adopted to deal with inefficiency in the procurement unit, unethical practices, noncompliance with SCM laws, and poor monitoring and evaluation mechanisms. The study contributes to the existing body of knowledge on the barriers facing the outsourcing of functions in the public sector.

Table of Contents

Declaration	i
Dedication	ii
Acknowledgement	iii
Abstract	iv
List of tables	vii
List of abbreviations	viii
CHAPTER ONE: INTRODUCTION AND OVERVIEW OF THE STUDY	1
1.1 Introduction	1
1.2 Background information on outsourcing and case study organisation	2
1.3 Statement of research problem	3
1.4 The research questions and objectives	4
1.5 Preliminary literature review	4
1.6 Summary of research methodology	5
1.7 The Significance, Justification and Contribution of the study	5
1.8 Limitations of the study	5
1.9 The structure of the dissertation	6
1.10 Conclusion	7
CHAPTER TWO: LITERATURE REVIEW ON THE CONCEPT OF OUTSOURCING	8
2.1 Introduction	8
2.2 A historical perspective on outsourcing	8
2.3 Theoretical models underpinning the concept of outsourcing	10
2.4 Emerging trends and factors driving outsourcing: Public Sector perspective	15
2.5 Outsourcing activities: A perspective from SASSA	18
2.6 Outsourcing challenges in the public sector	19
2.7 Process and procedure for outsourcing	22
2.8 Types of outsourcing	24
2.9 The dimensions of outsourcing	27
2.10 The classification of outsourcing	29
2.11 Conclusion	31

CHAPTER THREE: RESEARCH METHODOLOGY	32
3.1 Introduction	32
3.2 Research design and paradigms	32
3.3 Research methodology	33
3.4 The study sampling	34
3.5 The Reliability and Validity of a study	37
3.6 Conclusion	39
CHAPTER FOUR: DISCUSSION OF RESULTS AND INTERPRETATION OF FIN	DINGS 40
4.1 Introduction	40
4.2 Presentation and Data analysis	40
4.3 Response Rate	42
4.4 Test of the research objectives	42
4.5 Discussion of research findings	49
4.6 Conclusion	53
CHAPTER FIVE: SUMMARY, RECOMMENDATIONS AND CONCLUSION	54
5.1 Introduction	54
5.2 Summary of chapters	54
5.3 Summary of findings on research objectives	55
5.4 Study recommendations	56
5.5 Ethical considerations	58
5.6 Conclusion	58
REFERENCES	59
Appendix A: Interview Guide Questions	74
Appendix B: Informed Consent Form	75
Appendix C: Ethical Clearance	76
Appendix D: Turnitin report	77

List of Tables

Table 4.1 Demographics of respondents of the Face-to-Face interview session	. 40
Table 4.2 Demographics of respondents of the Focus Group discussions	. 40

List of Abbreviations

3PL Third Party Logistics

BOP Business Outsourcing Process

B-BBEE Broad-Based Black Economic Empowerment

COSATU Congress of South African Trade Unions

CCT Core Competency Theory

CM Contract Management

CPS Cash Pay Master

GDP Gross Domestic Product
IT Information Technology

SASSA South African Social Security Agency

SAPO South African Post Office

PPPFA Preferential Procurement Policy Framework Act

PFMA Public Finance Management Act

R-BT Resource-Based Theory

SONA State of the Nation Address

Chapter One

Introduction and Overview of the Study

1.1 Introduction to the study

The South African government has been challenged by irregularities and fruitless expenditure in public sector outsourcing over the past decades (National Treasury, 2019). The National Treasury has relentlessly called for cost-cutting measures and the need to increase operational efficiency in public entities (National Treasury, 2019). Traditionally, outsourcing in the public sector has mainly focused on minor activities such as travelling, maintenance, catering, cleaning and security services. However, recent developments have seen activities such as marketing, supply chain management, human resource management, information technology, and others, being outsourced. In developed nations such as the United Kingdom, United States of America, Australia and New Zealand, public sector outsourcing plays a vital role in improving organisation performance and reducing operating cost ((Dhall 2013; Harland et al., 2005 and Hien, 2005). Though it is well-established that outsourcing can help in overcoming various developmental challenges, there have been concerns in South Africa over outsourcing in relation to cost-cutting measures, as well as achieving developmental goals. Some schools of thought believe that the inability of outsourcing to address the national developmental challenges is attributed to various challenges associated with outsourcing. For example, the Congress of South African Trade Unions (COSATU) (2015), argues that public sector outsourcing does not reduce an organisation's operating costs but is aimed at profit maximization for the owner. In support of this, the South African 2019 State of the Nation Address (SONA) noted that public sector expenditure is expected to be reduced by R26.1 billion over the next three years through cutting non-core services (National Treasury, 2019). Literature has documented various challenges facing public sector outsourcing which include amongst others, organisational politics, employee morale, corruption and price inflation (Rispel, 2015; Jeppesen, 2015 and Osei-Tutu et al., 2010).

Given the above, the main focus of this study is to identify the barriers facing the outsourcing of functions in a South African public sector organisation. The identified department is the South African Social Security Agency (SASSA). This chapter presents the background to the study, a statement of the research problem, the research questions, and objectives. In addition, the chapter discusses a preliminary literature review, and summarises the research methodology, significance, justification and contribution of the study. Furthermore, limitations are described and an outline of the rest of the chapters as well as the summary of the chapter is presented.

1.2. Background information on outsourcing and the case study organisation

Outsourcing of organisational functions in the public sector has been a trend over the years (Badenhorst-Weiss, 2012). Productivity and real gross domestic product (GDP) per capita have declined since 2015 (South African Reserve Bank, 2018; Kreuser and Newman, 2018). Moreover, South Africa has low economic growth which entrenches inequality and poverty (Aterido, Hlatshwayo, Pieterse and Steenkamp, 2019). This has triggered an increasing need for the South African government to cut down on spending (National Treasury, 2019). Arrowsmith (2010) argues that poor management, unethical practices, and lack of control mechanisms have caused major operational challenges and have resulted in budget cuts. The 2019 Budget confronts these challenges by revisiting public sector spending plans and reducing outsourcing of non-core activities, amongst others (National Treasury, 2019). The National Treasury (2019) announced a re-positioning of public entities to minimize wastage in daily operations. As noted earlier in Section 1.1, this study focuses on outsourcing practices in the South African Social Security Agency (SASSA).

The agency, SASSA, is mandated to ensure the provision of comprehensive social security services for the vulnerable and poor citizens of South Africa (SASSA, 2004). The objective of SASSA is to act, as the sole agent, to ensure the efficient and effective management, administration and payment of social assistance, as well as to serve as an institution to manage broader social security benefits within the constitutional and legislative framework (SASSA, 2015). The legislative mandates for SASSA are the Social Assistance Act (2004) and the South African Social Security Agency Act (2004). The Social Assistance Act (2004) is a national legislative framework for the provision of different types of social grants, social relief of distress, the delivery of social assistance grants by a national Agency and the establishment of an Inspectorate for Social Security, whilst the South African Social Security Agency Act, 2004, provides for the establishment of the South African Social Security Agency as a schedule 3A public entity in terms of the PFMA. The principle aim of the Act is to make provision for the effective management, administration and payment of social assistance and service through the establishment of the South African Social Security Agency. SASSA has its departments at provincial level, located in the nine provinces of South Africa.

Unique to other public sector entities, SASSA is facing challenges on outsourcing processes. Such challenges include but are not limited to price inflation, organisational politics, corruption and inadequate monitoring and control of suppliers (SASSA, 2018). SASSA outsources various activities, such as informing beneficiaries about the benefits of a new card, persuading beneficiaries to switch to the new card, and reminding beneficiaries about services offered by the agency. SASSA outsources their advertisement of tenders and vacancies to local newspapers in order to access high numbers of potential applicants and potential service providers (SASSA, 2012).

Currently, the service provider is charging R10 per head for the payment of a social grant (Cash Payment Services, 2017). Due to the challenges facing the department, SASSA decided to standardize operations by selecting one service provider to distribute pension grants in all nine provinces, which led to the appointment of Cash Pay Master Services (Overseas Development Institute (ODI), 2011). According to ODI (2011), the main objectives were to retain control over the service provider while reducing handling costs, and the creation of one standardized grant recipient database. However, Cash Pay Master Services illegally processed value-added services such as loans, bill payments and funeral policies, which was against the initial agreement between SASSA and CPS (Net1, 2015). As a result, SASSA dropped the services of CPS and outsourced payment services to the South African Post Office Bank (South African Post Office, 2018). The new contract between SASSA and the South African Post Office (SAPO) was signed with the focus on reducing costs and improving the quality of payment services, while protecting social grant recipients from loan sharks and unauthorized deductions (Department of Social Development, 2018).

The public sector has long outsourced activities, such as promotion and advertising campaigns (McGovern and Quelch, 2010). The outsourcing of activities has been considered a general practice in the public sector, which presumably allows organisations to focus on core competencies, increase customer service and reduce costs. The State of the Nation Address (2019) budget emphasizes the need to cut costs, especially those associated with outsourcing.

The outsourcing process requires consideration of costs, risks, quality, expertise, nature of demand and available capacity (Fawcett, Ellram and Ogden, 2012). As mentioned earlier in Section 1.1, this study seeks to identify the barriers facing the outsourcing of functions in the South African Social Security Agency at the Umzumbe local office in Port Shepstone. Umzumbe is a part of the Umzumbe Municipality and the local office is located in Umtwalume, a rural area of Port Shepstone on the South coast of KwaZulu-Natal (Statistics South Africa, 2013). The study specifically examined barriers to the effective use of outsourcing in the South African Social Security Agency.

1.3. Statement of research problem

According to the National Treasury, South African public entities have embraced the outsourcing concept but still struggle with achieving the benefits of outsourcing due to corruption, price inflation, organisational politics and recession (National Treasury, 2016). In addition to these challenges, the South African Social Security Agency (SASSA) at the national level experiences more challenges in the areas of monitoring and controlling service providers in all provinces. Such a challenge is evident in the payment of varying amount of social grants across the provinces, as pointed out in Section 1.2. The challenge associated with outsourcing has led to the government's continual effort to reduce

operational costs in the public sector. As a result, public entities have continued to implement various measures to contain operational costs and eliminate all non-essential expenditure (National Treasury, 2018). The reason behind government's outsourcing of functions is to reduce operational costs and obtain more specialist skills from service providers (Fawcett *et al*, 2007). The outsourcing of functions in the public sector has been considered a major concern for the effective and efficient running of organisations. Lately, government is realizing that the cost-cutting advantage of outsourcing is no longer being achieved (National Treasury, 2016). In view of the outsourcing challenges facing SASSA, this study addresses the question: What are the barriers affecting outsourcing functions in the South African public sector? Thus, it is anticipated that this study will assist SASSA and other public institutions to eliminate barriers associated with outsourcing while improving organisation efficiency and effectiveness.

1.4. The research questions

In view of the outsourcing challenges facing SASSA, this study addresses the question: What are the barriers affecting outsourcing functions in the South African public sector? Furthermore, secondary research questions derived from the main research question are:

- What are the barriers facing outsourcing of activities in the South African public sector?
- To what extent do challenges of outsourcing affect the organisation's operations?
- What are the monitoring measures in place and how do they affect the outsourcing activities in the South African public sector?

1.4.1. The aim and objectives of the study

The main aim of the study is to identify the barriers affecting outsourcing functions in the South African public sector. Subsequent research objectives emanating from the aim are:

- To investigate the barriers facing outsourcing functions in the South African public sector;
- To determine the extent that outsourcing-related challenges impact on the organisation's operations;
- To examine the existing monitoring measures and their impact on the outsourcing functions in the South African public sector.

1.5. Preliminary literature review

Pauw (2011) argues that cost reduction benefits of outsourcing activities in the South African public sector are not achieved due to high costs, a lack of adequate mandatory service standards and a lack of monitoring mechanisms for service providers. In addition, the public sector faces other challenges in outsourcing which include incompetence, a lack of accountability, corruption and the negligence of public servants (De Lange, 2011; Acevedo, Rivera, Lima and Hwang, 2010 and Osei-Tutu, Badu and

Owusu-Manu, 2010). The study adopted two theories deemed relevant to the research focus area, namely Relational View Theory and Core Competency Theory.

1.6. Summary of research methodology

The design of this study is of an exploratory nature in which the researcher used a qualitative approach and a non-probability purposive sampling method. The researcher adopted a purposive sampling method because of its advantages of getting relevant information and saving effort, money and time. This research used face-to-face interviews and focus group discussions. The sample comprised 41 employees from the South African Social Security Agency, Umzumbe Local Office. The primary data was collected using in-depth face-to-face interviews with 6 senior government officials and focus group discussions involved 35 discussants. Data was analysed using content analysis. Interview guided questions were constructed in line with the research topic and objectives of the study to ensure the reliability and validity of the study.

1.7. The significance, justification and contribution of the study

Research into outsourcing has been extensively studied by professionals and academic (Eyaa and Oluka, 2011). Research into outsourcing within the South African public sector is very important because of its strategic role in achieving transformation and socio-economic development. Although there are a growing number of studies in the area of outsourcing, there seem to be minimal studies focusing on the outsourcing of functions in the public sector in South Africa, more specifically in the bigger public sector entities such as the South African Social Security Agency. Therefore, the study contributes to a better understanding of the barriers affecting outsourcing functions in the public sector. From an academic point of view, the study contributes to further research in the context of outsourcing of activities.

1.8. Limitations of the study

The study focused on one SASSA local office in Umzumbe and findings are limited to the focus area and cannot be generalised outside the study area. Given that the study was conducted in only one of the SASSA local offices, the researcher feels that the incorporation of more SASSA offices would have provided more in-depth information about the research problem. As such, the researcher considers this a limitation for the study. Due to the limited office space provided for the focus group discussions, the researcher had to divide them into 3 groups, which added to the estimated time for data collection and increased travelling costs, though with relatively no negative impact on the quality of data collected.

1.9. The structure of the dissertation

The structure of the dissertation provides information on the layout of the content or chapters presented in the dissertation. This dissertation comprises of six chapters as follows:

Chapter One: Introduction and overview of the study

Chapter one provides an introduction and background to the study, a statement of the research problem, the research questions and objectives. Moreover, the chapter provides preliminary literature review, and a summary of the research methodology, significance, justification, contribution and limitations. In addition, the chapter discusses the limitations of the study, and the structure of each chapter.

Chapter Two: Review of relevant literature on outsourcing

Chapter two reviews current literature on the concept of outsourcing functions in the public sector. This chapter explores existing literature and provides in-depth knowledge on the area of outsourcing. The information covers a historical perspective on outsourcing, theoretical models underpinning the concept of outsourcing and emerging trends and factors driving outsourcing in the public sector context. In addition, this chapter addresses outsourcing activities, a perspective from SASSA, and outsourcing challenges in a public sector organisation. Further, this chapter discusses processes and procedures for outsourcing, dimensions, classification of outsourcing and a conclusion.

Chapter Three: Research methodology

Chapter three addresses the research methodology used in the data collection of the study. The chapter covers the introduction, research design and paradigms, approaches to the research study and research instruments adopted for the study. Finally, the chapter concludes by outlining the research population, sampling technique used in the study, data collection, analysis methods, reliability and validity of the study.

Chapter Four: Data analysis and discussion of findings

Chapter four covers the data presentation and analysis. It provide an introduction, presentation and analysis of data, response rate and the test of the research objectives. In addition, this chapter discusses the research findings.

Chapter Five: Conclusion and recommendations

Chapter five draws a conclusion and makes recommendations. The chapter provides a summary of all the chapters, and a summary of the research findings. In addition, this chapter discusses the study's recommendations and ethical considerations, and lastly, draws to a conclusion.

1.10. Conclusion

The outsourcing processes in the public sector have taken centre stage amongst professionals, academics and all stakeholders concerned due to its ineffectiveness in addressing the developmental challenges facing the nation. South African public sector outsourcing practices are characterised by, amongst others, corruption and inadequate control measures, resulting in the nation struggling with achieving various political and national mandates. This chapter provides information on the background to the study, a statement of the research problem, the research questions and objectives. Moreover, the chapter discusses a preliminary literature review, summary of the research methodology, significance, justification, contribution and limitations. In addition, the chapter discusses the limitations and provides a summary of each chapter in the dissertation. The next chapter, chapter two, presents information related to the general concept of outsourcing.

Chapter Two

Literature Review on the Concept of Outsourcing

2.1. Introduction

Outsourcing is a topic of debate amongst professionals, academics and various stakeholders (Turley and Perera, 2014). The concept has been embraced by many organisations in the quest to achieve competency and cost saving (Sani, Dezdar and Ainin (2013). The achievement of the benefits of outsourcing in the public sector has been a challenge for governments, mostly from developing countries. In South Africa, outsourcing is used as a mechanism and strategy to enable the government to implement transformation policies and socio-economic development (Eyaa and Oluka, 2011). South Africa, like other nations, still faces challenges associated with outsourcing, which include but are not limited to, corruption, incompetence and negligence, resulting in the wastage of public resources. This chapter reviews existing literature on the concept of outsourcing. It provides an historical perspective on outsourcing and theoretical models guiding the concepts of outsourcing, which includes the definition of outsourcing and information on the theoretical models adopted for the study, namely Relational View Theory and Core Competency Theory. Moreover, this chapter reviews emerging trends and factors driving outsourcing in the public sector, outsourcing challenges in a public sector organisation, and processes and procedures for outsourcing. Thereafter, the chapter reviews types of outsourcing, and dimensions and classifications of outsourcing before concluding the chapter.

2.2. A historical perspective of outsourcing

Literature has documented different perspectives on the emergence of the concept of outsourcing (O'Sullivan, 2013). While some scholars have traced outsourcing to the early 1950s, others posit that outsourcing emerged later due to changes in the business environment. Brewster *et al.* (2002) argue that the outsourcing phenomenon emerged between the 1950s and 1960s, a period when businesses started using diversification strategies to gain competitive advantages. According to Willcocks and Lacity (1995), outsourcing research emerged during the late 1980s to the early 1990s. Furthermore, to some scholars the outsourcing phenomenon gained importance in the last 20 years due to globalization and transformation (Potrafke, 2018; Pensiero, 2017; Jenson, 2007; Corbet, 2004). In the 1970s, outsourcing faced many challenges as organisations attempted to compete in the global market but failed because of a lack of agility that resulted from bloated management structures (Brewster *et al.*, 2002). Due to the rapid growth of technology and software, innovative models of outsourcing came about in the late 1980s and 1990s. In the early 1990s most organisations increased their flexibility and creativity by developing new strategies that helped organisations to focus on their core business (Handfield, 2006). According to McKee (2000), in the 20th century, the public sector focused on outsourcing most of its activities. In

the 21st century, organisations experienced greater improvements in technology, communication and innovation, which made it easier for outsourcing to be successful.

There are many perspectives on the understanding and meaning of outsourcing. Outsourcing is defined as a process whereby a firm no longer wants to pursue activities internally and so acquires them from external suppliers (Ellram and Billington, 2001; Gilley and Rashood, 2000; Quinn and Hilmer, 1994). From another scholarly point of view, outsourcing is a process of hiring and managing a contractual relationship that had been previously in-sourced (Momme, 2001). Outsourcing can be defined as a process of transferring activities that are done internally to an external supplier (Schialfino and Berkowitz, 2015; Iqbal and Dad, 2013). Outsourcing occurs when an organisation contracts with another organisation to offer services of a major activity or function which may include manpower, skills, technology, knowledge, or services at a certain period and price (Belcourt, 2006; Fawcett et al., 2007). The outside firms providing the outsourcing services are third-party providers, called service providers (Bera, 2017). Outsourcing does not mean the abandoning of responsibilities from the parent company to the service provider (Jensen and Stonewash, 2004). Rather, the parent company oversees the service provider. Outsourcing helps organisations to identify critical and non-critical processes that can be outsourced (Handfield, 2006). As such, both parties in outsourcing deals do not share the profits and efficiency contribution (Handfield, 2006). According to Kakabadse and Kakabadse (2000), outsourcing is very important in organisational strategies. Clear outsourcing strategies help organisations to achieve efficiency and reduce costs (Feeny, Lacity and Willcocks, 2005). Lately, organisations are focusing on outsourcing with more emphasis on cost-saving. There are many reasons for government to outsource functions rather than to in-source, which include specialized skills, better quality services, knowledge and competencies. Sometimes, the public sector relies on the private sector as government cannot be an expert in delivering all types of services. Outsourcing within the public sector is seen as a business strategy in which public entities strive to minimize costs and focus on core competencies, as well as improving the quality of services (National Treasury, 2018; Belcourt, 2006; Bhatia and Drew, 2006).

In South Africa, the use of outsourcing has been growing rapidly. Some studies have shown a positive impact, such as increases in efficiency and cost savings, while others show negative impacts resulting from contracting out and tendering (Bester, 2000). According to the National Treasury (2017), government departments need to minimize fruitless and wasteful expenditure through cutting operational costs, combating waste, eradicating corruption and increasing inefficiency. According to Sani, Dezdar and Ainin (2013), the outsourcing of functions in the public sector is increasingly becoming a strategic necessity. However, there are some indications that these benefits are not being realized. Lately, the public sector is realizing that the cost-cutting advantage of outsourcing is no longer

being achieved, rather, organisations are incurring debt because of outsourcing. In South Africa, public sector outsourcing is fully guided by the Preferential Procurement Policy Framework Act and supported by the National Treasury (Ambe and Badenhorst-Wiess: 2012; National Treasury, 2005). Public sector outsourcing ensures the empowerment of Small, Medium and Micro Enterprises through Broad-Based Black Economic Empowerment. In addition, outsourcing is firmly rooted in the reduction of costs and gaining competitive advantage.

2.3. Theoretical models guiding the concept of outsourcing

The term "theory" can be defined as a hunch, idea, framework, concept or model that can help one to understand a topic of interest, describe a phenomenon or solve a problem (Sekaran and Bougie, 2016). Similarly, frameworks are strategies of inquiry that brings together claims being made on what constitutes knowledge (Creswell, 2014). Researchers have defined theoretical models from different perspectives but depicting similar contexts in its meanings (Tidwell, 2012; Sinclar, 2007; Widd and Diggines, 2010). Sinclair (2007) defines a theoretical model as a map or plan which helps a researcher to seek as much knowledge as possible using information from previous studies. Wiid and Diggines (2010) consider theoretical models as a group of ideas used to examine a topic from different viewpoints in order to seek answers on research questions. In essence, a theoretical model tests a theory using instruments and helps researchers to understand, predict, explain, and challenge a phenomenon or existing knowledge, thereby representing a researcher's beliefs in a relationship of certain concepts (Sekaran and Bougie, 2016; Sinclair, 2014; Creswell, 2014). A theoretical model plays an important role in a research process, providing support and a strong scientific research base to a dissertation (Wiid and Diggines, 2010).

2.3.1. Theoretical models adopted for the study

This study is guided by a theoretical and a legislative framework. The use of both a theoretical and a legislative framework stemmed from the researcher's quest to understand the nature of outsourcing activities and the successful management of outsourcing. The researcher believed that the use of both frameworks will provide a better understanding of the concept of outsourcing. For the purposes of this study, two models of outsourcing were adopted, namely Relational View Theory and Core Competency Theory.

According to Kraaijenbrink *et al.*, (2010), the Relational View Theory (RVT)/ Resource-Based View Theory (RBV), has been one of the dominant theories in strategic management since the 1990s. The Relational View Theory (RVT)/ Resource-Based View Theory (RBV) objective is to create and maintain a sustainable competitive advantage throughout the industry (Barreto, 2010). Furthermore, competitive advantages of an organisation can be gained through organisational capabilities, resources

and competencies (Diez-Vial and Alvarez-Suescun, 2010). Organisational resources have an impact on management capabilities, which in turn are the sources of core competencies that ultimately lead to a competitive advantage (Kraaijenbrink et al., 2010; Barney and Hesterly, 1996). This means that competitive advantages are gained from internal resources as well as external resources through relational networks (Kraaijenbrink et al., 2010). According to Kraaijenbrink et al., (2010), RBV rests on two key assumptions: resources are different from any other organisation and may not be easily copied by other organisations. Moreover, the RBV theory has to be deployed in an appropriate manner in order to create a competitive advantage: to be unique from competitors; valuable in order to create a marketing positioning; not easily copied by competitors; and not easily substituted by customers (Barney et al., 2011; Kraaijenbrink et al., 2010; Diez-Vial and Alvarez-Suescun, 2010; Barney, 1991). Thus, the Resource-Based View in outsourcing builds from a proposition that an organisation that lacks valuable, rare, inimitable and organised resources (VRIO) will not survive in the global market (Kraaijenbrink et al., 2010; Kaplan and Norton, 2004). The VRIO proposition addresses the exploitation of resources by the organisation to provide competitive advantages that cannot be easily copied by competitors. Therefore, the most prominent use of RVT/RBV is in the preparation phase of outsourcing processes, which compels the business to create a sustainable competitive advantage and transformation in a strategic manner (Diez-Vial and Alvarez-Suescun, 2010; Kaplan and Norton, 2005). Furthermore, RBV theory is applied in the procurement process to check the eligibility of a supplier (Kutsikos and Mentzas, 2011; Graf and Mudambi, 2005). RBV is a viable option for outsourcing when there is no strategic need to in-source a product or component (Kraaijenbrink et al., 2010; Kaplan and Norton, 2005 and Barney, 1991). Moreover, a combination of RBT and TCE theories is appropriate for the interpretation of outsourced results (Nordigarden et al., 2014; Barthélemy and Quélin, 2006).

Similarly, the concept of Core Competency Theory (CCT) is developed alongside the Resource-Based View. According to Nasiopoulos, Sakas and Vlachos (2014), CCT is used as a tool to determine the success of an agreement. Furthermore, CCT is a combination of different skills which integrate various technological methods and analyse vendor competencies (Nasiopoulos *et al.*, 2014; Mehta *et al.*, 2006). Moreover, core competencies are defined as the ability of organisations to develop products or services that are not easily imitated (Fawcett *et al.*, 2012). In addition, core competencies can be achieved through understanding customer needs and developing products or services that will exceed those needs (Nasiopoulos *et al.*, 2014; Belcourt, 2006).

2.3.2. International frameworks guiding outsourcing

According to McIvor (2000), the outsourcing frameworks were developed to overcome the weaknesses of outsourcing. Sinclair (2007) stated that frameworks are formulated to develop, explain, predict or plan and examine a topic. According to McIvor (2000), a good outsourcing framework should provide

for an acceptable value chain perspective, core competency in thinking, a measure for performance, and cover an acceptable strategic context.

2.3.2.1. Value chain perspective

Value chain analysis is a process of dividing various business functions into primary and supporting activities, analysing business activities and assessing how to create competitive advantage (Ambe and Badenhorst-Weiss, 2012, 2016). According to Fawcett *et al.* (2012), international organisations use value chain analysis as a tool for identifying activities or functional areas within a firm that can be outsourced. Moreover, the value chain analysis helps international organisations to increase efficiency and review logistics and supply chains to create or maintain competitive advantages (Fawcett *et al.*, 2012).

2.3.2.2. Core competency

One of the outsourcing objectives in an organisation is to focus on core competencies (Fawcett *et al.*, 2012). Organisations within the United Nations (UN) identify service providers that best meet the organisation's needs, with the aim of increasing economies of scale, flexibility and improving organisational efficiency (UN, 2012). According to Fawcett *et al.* (2012), successful outsourcing allows organisations to concentrate on carrying out their core competencies, which usually leads to the efficient management of the efficiency and human resources of an organisation. Moreover, the outsourcing of non-core functions provides organisations with flexibility to re-direct resources to critical organisation functions (Fawcett *et al.*, 2012).

2.3.2.3. Performance measurement

Effective policy-making in outsourcing requires information on whether the organisation is achieving the intended results (Acevedo, Rivera, Lima and Hwang, 2010). Strong performance management systems help organisations to identify strengths, weaknesses and accountability (Acevedo *et al.*, 2010). The use of performance management ensures efficient, economical procurement, whether information should be recorded, and non-performance should be reported in information systems (Stemele, 2009).

2.3.2.4. Strategic context

According to the UN (2012), while outsourcing addresses cost-saving measures, it should also provide room for fairness, transparency and integrity through competition, best value for money, economy and effectiveness.

• **Fairness:** The UN posits that competition for outsourcing should be conducted in a fair manner that assures reasonable administrative procedures. This is to ensure that all interested parties

can engage in fair and competitive tendering processes (UN, 2012). It is important for an organisation to ensure impartial treatment and fairness to all business partners and stakeholders, making the organisation's procurement free of bias and favouritism (Luyt, 2008).

- Transparency and integrity through competition: The UN emphasizes the need for organisations to comply with ethical procurement standards in order to eliminate fraud and corruption, avoid accepting gifts and deal with conflicts of interest with potential suppliers and current suppliers (Jeppesen, 2010). Transparency means keeping stakeholders informed on decisions taken and operational plans (De Lange, 2011). According to Jeppesen (2010), transparency allows diverse views and multiple perspectives to influence policy decisions in an organisation.
- Best value for money: Best value for money refers to selection offers that best suit the strategic objectives of outsourcing such as social and environmental costs, quality or other strategic objectives of the buyer (Ambe and Badenhorst-Weiss, 2012). According to Watermeyer (2011), price alone is not a realistic indicator and organisations look beyond the price factor because cheaper prices that meet mandatory requirements do not necessarily mean a good deal. To achieve best value outcomes over the procurement cycle, it is important to consider costs and benefits. The best value for money can be achieved internally through continuous improvement in internal systems and processes; avoidance of delays and unnecessary costs with suppliers; and controlling and monitoring supply arrangements to ensure that they provide the expected benefits (Pauw and Wolvaardt, 2009; Bolton, 2006).
- Economy and effectiveness: Economy and effectiveness means providing a suitable solution to the organisation's need with regard to the right price, quantity and quality (Watermeyer, 2011). According to Bolton (2003), effectiveness ensures that the interests of the end-user is met, while economy protects the organisation's budget.

2.3.3. Legislations underpinning outsourcing in South Africa

Legislation means "a set of rules" set by parliament and it has to be obeyed by everyone (Western Australia Parliamentary Counsel Office, 2011:2). Legislation is an important tool used by government in protecting and guiding society (Tukuta *et al.*, 2015). Adhering to legislation to some degree ensures value for money; open and effective competition; ethics and fair practice; reporting and accountability; and equity (National Treasury, 2017). The importance of legislation in outsourcing cannot be overemphasised. Legislation provides a structure of authority, smooth working, and fulfilment of the constitutional mandates of the public sector (Watermeyer, 2011).

In dissertation writing, legislation provides the reader with the information on legislative regulations related to the research focus area. In South Africa, the National Treasury provides all relevant policies addressing the concept of outsourcing with the aim of regulating how the public sectors operate in outsourcing processes (Kingpin, 2015). Moreover, outsourcing in the public sector is used to fulfil constitutional obligations through providing a fair and transparent procurement process (Turley and Perera, 2014). According to Eyaa and Oluka (2011), the public sector uses outsourcing for the socioeconomic upliftment of the citizens. South African legislative frameworks for outsourcing are the Constitution of South Africa; Municipal Finance Management Act No 56 of 2003; Public Finance Management Act No 1 of 1999; Preferential Procurement Policy Framework Act No 5 of 2000; Broad-Based Black Economic Empowerment Act No 53 of 2003; Prevention and Combating of Corrupt Activities Act No 12 of 2004; Promotion of Administrative Justice Act No 3 of 2000; and the Promotion of Equality and Prevention of Unfair Discrimination Act No 4 of 2000.

The primary objective of the Republic South Africa's Constitution, Act 108 of 1996: Section 217(1) is to ensure that outsourcing practices and processes in all of the public sector are cost-effective, unbiased, fair, transparent, competitive and accessible to all of the public (Watermeyer, 2011). The secondary objectives of the South African Constitution are to redress the mistakes of apartheid and implement transformation policies that are aimed at socio-economic development (Turley and Perera, 2014).

The purpose of the Municipal Finance Management Act (MFMA) No 56 of 2003 is to establish a regulatory framework for SCM, which includes outsourcing/procurement in municipalities and municipal entities (Watermeyer, 2011; National Treasury, 2003).

Similarly, the purpose of the Public Finance Management Act (PFMA) No 1 of 1999 legislation is to regulate public sector management of funds, minimize wastage of funds, and enforce lawful practices of moral efficiency management to minimise the mismanagement of limited public resources (National Treasury, 2005).

The Preferential Procurement Policy Framework Act (PPPFA) No 5 of 2000 aims to establish the manner in which the preferential procurement rules and regulations are to be implemented in the public sector. Moreover, the PPPFA ensures that all public entities need to follow standard criteria when evaluating suppliers and establishes the manner in which preferential procurement policies are to be implemented (National Treasury, 2017).

The Broad-Based Black Economic Empowerment Act (BBBEE) 53 of 2003 ensures that businesses owned by previously disadvantaged people have an equal chance of participating in the South African economy, such as Blacks, Coloureds, Indians and Chinese (National Treasury, 2005). The government of South Africa uses outsourcing in the implementation of transformation policies and redress of unfair practices of the past. Such policies and practices are evident in the Broad-Based Black Economic Empowerment (BBBEE) Act No 53 of 2003. The BBBEE Act ensures that businesses owned by marginalised groups, including women, have equal opportunity to participate in the country's economic activity (National Treasury, 2017). According to Van Weele (2005), BBBEE is a strategy aimed at redressing the exclusion of black people and women from participating in the South African economy, and supporting skill development, income equality and equal opportunities.

The Prevention and Combating of Corrupt Activities Act (PCCAA) No 12 of 2004 ensures that government criminalises corruption, fraud and other related activities as an offence (National Treasury, 2003). Furthermore, this Act restricts individuals or enterprises that are convicted of corruption, fraud and other unethical practices related to public sector contracts and tenders. Moreover, the PCCAA's duty is to ensure accountability and encourage whistleblowing on corrupt transactions related to procurement activities in the public sector (Ambe and Badenhorst-Weiss, 2012).

The primary objective of the Promotion of Administrative Justice Act (PAJA) No 3 of 2000 is to ensure that administration action is properly documented, procedurally fair and reasonable and lawful (National Treasury, 2003). Furthermore, PAJA imposes a duty on the state to ensure fair administrative procedures that permit those affected by unfair administrative action to request reasons for such administrative actions and remedies proceeding for judicial review (Bolton, 2003).

The Promotion of Equality and Prevention of Unfair Discrimination Act No 4 of 2000 prohibits unfair discrimination on the basis of gender or race by any person or organisation through the denial of access to contractual opportunities (National Treasury, 2005).

2.4. Emerging trends and factors driving outsourcing: A public sector perspective

Outsourcing is one of the most influential trends re-shaping management (Gurría, 2016; Uyarra and Flanagan, 2009; Kashap, 2004). According to Mol and Kotabe (2011), the process of outsourcing, that is, the organisational relocation of activities or entire functions outside a firm's boundaries, has been a key driver of business transformation over the past decades. Studies have shown various reasons for firms' outsourcing practices. For example, Andersson, Fransson and Rehme (2006) identified five analysis areas for outsourcing decisions which form the continuum between Transactions cost and

Resource-based theory. The identified areas by Michela and Carlotta (2011) are cost consequences, operational/capacity constraints, core competency, access to external resources, and financial motives.

2.4.1. Cost consequences

Outsourcing success depends on a supplier's ability to deliver service safely and timeously. The core consequences of outsourcing occur in the case where service providers fail to deliver services on time, which can affect the sales and reputation of the organisation (Iqbal and Dad, 2013). Kingpin (2015) argues that outsourcing processes do not provide solutions for every resource shortage, as there are situations where outsourcing is not a good idea. Core consequences of outsourcing include the case where costs of the service provider exceed the organisation's budget. Moreover, outsourcing production may lead to customers, citizens and communities boycotting the product. Furthermore, outsourcing in the public sector reduces public employment because the state relies on private contractors (Petersen, Hjelmaa and Vrangback, 2018). Despite the ever-growing outsourcing phenomenon in the public sector, it is difficult to come up with strategies that reduce problems and challenges facing public sector outsourcing (Choy *et al.*, 2005).

2.4.2. Operational or capacity constraints

Outsourcing is considered in the absence of competence, while insourcing is appropriate when there is an availability of capacity. There are uncontrollable elements when the business outsources globally. This includes the legal systems and politics of foreign markets, global economic conditions, social and cultural customs, and technological forces (Hugo, Baden-Horst-Wiess and Van Billon, 2004).

2.4.3. Core competency

Core competencies refer to a business strategic advantages/ main strengths of an organisation (Fawcett *et al.*, 2012). Moreover, the reasons for outsourcing stem from a company's quest to focus on the core business and cost-cutting (Farrell, 2010). According to Feeny *et al.* (2005), understanding customer needs helps organisations to align core competencies to best meet various organisational goals such as performance improvement and faster services.

2.4.4. Access to external resources

One of the advantages of outsourcing is accessing advanced technologies and highly skilled labour (Fawcett *et al.*, 2012). Moreover, access to an experienced service provider usually leads to better quality, prompt and faster delivery (Wongleedee, 2016). According to Poutvaara (2014), outsourcing is ideal when the organisation does not possess the necessary skills and technology to fulfil its mandate.

2.4.5. Financial motives

Organisations who master the art of outsourcing are achieving cost reduction in daily operations (McIvor, 2005, Feeny *et al.*, 2005). Private sector organisations outsource different functions with the aim of increasing market share, accessing supplier's skills, exploiting business opportunities offered by the network of suppliers, and to focus on internal investments (Vintar, 2011). One of the outsourcing objectives in the South African public sector has been about minimizing wastage on government procurement (Belcourt, 2006; Kumar and Eickhoff, 2005; Kakabadse and Kakabadse, 2002; Kremic *et al.*, 2006). The public sector is not profit oriented and pricing is set by government regulators; usually prices are charged to break even, where the price is set to equal average costs (Pessoa, 2009). According to Bhatia and Drew (2006), organisations use outsourcing as a cost-cutting measure without considering the impact of the outsourced service on the overall process flow. Moreover, outsourcing presents financial implications to a parent organisation, which often leads to termination of agreements between both parties (Kavcic and Tavcar, 2008). Furthermore, the concept of cost savings has been a challenging factor facing outsourcing in the public sector (Arrowsmith, 2010).

2.4.6. Strategic tool

Outsourcing activity plays a vital role in enhancing an organisation's overall strategy (Dinu, 2015; McIvor, 2005). Organisational decisions to outsource or in-source can be referred to as a make or buy decision (Fawcett *et al.*, 2012). Some considerations regarding outsourcing includes: costs, risks, availability, capacity, expertise, quality considerations and nature of demand (Fawcett *et al.*, 2012). According to Feeny *et al.* (2005), outsourcing is built around competitive advantages and cost reduction. In addition, some organisations may decide to use outsourcing to improve on performance, access low-cost labour, and benefit from outside expertise (Fawcett *et al.*, 2012). Outsourcing in some organisations is used as a strategy to get rid of an underperforming department, thus reducing the number of employees (Fawcett *et al.*, 2012). The parent organisation decides which part of the business function needs to be insourced and what can be outsourced (Wiesel, 2015). Outsourcing with a clear cut strategy yields good results for an organisation (McIvor, 2005; Feeny *et al.*, 2005). Furthermore, outsourcing leads to cost reduction through increased efficiency and lower labour costs resulting from an outsourced function (Kingpin, 2015). Denning (2013) argues that outsourcing does not always increase efficiency and lower costs, rather it increases wastage. Thus, the need to manage an organisation's resources effectively remains a challenge for organisational activities.

2.4.7. Socio-economic development purposes

Outsourcing in any organisation has to consider socio-economic developments, such as the well-being of the community and management of the environment (Bedy, Eklund and Najafi, 2008). In addition, outsourcing in the South African public sector has shifted drastically from cost saving to mechanisation

and has become a strategic tool enabling government to implement transformation policies and socio-economic development, redress unfair practices of the past and promote ecological accountabilities (Turley and Perera, 2014; Eyaa and Oluka, 2011; Watermeyer, 2011; Bolton, 2006). Moreover, according to Mahmood (2010), the South African public sector spends billions of Rands on outsourcing, with the aim of addressing certain socio-economic disparities of the past, through promoting the empowerment of Small, Medium and Micro Enterprises (SMMEs).

2.4.8. Quality improvement

According to Kruger *et al.* (2007), quality is defined as meeting customers' expectations. Quality is regarded as a major driver of competitive strategy in any industry and it can increase organisational revenue, improve productivity and reduce costs (Kingpin, 2015). To achieve quality improvement, organisations and stakeholders need to understand various concepts of quality and implement those concepts throughout the organisation. In recent years, the public sector has outsourced different functions with the aim of improving service and quality attributes such as empathy, responsiveness and building a good reputation with customers (McIvor, 2005; Feeny *et al.*, 2005). Moreover, the public sector outsources different functions with the objective of improving quality and efficiency (Iqbal and Dad, 2015; Cook, 2014).

2.4.9. Business efficiency

According to Iqbal and Dad (2013), the objectives of outsourcing range and include improving operational efficiency, increasing returns on shorter investment cycles, accessing experience, innovative suppliers and modern technology, and the minimization of risk, improved control and better governance. Moreover, private provision tends to be more efficient than public provision due to competitiveness in the private sector (Poutvaara, 2014). Furthermore, organisational decisions to outsource may be affected by various factors such as cost savings, capital conservation, reliability, Broad-Based Black Economic Empowerment (BBBEE), improved services and organisational politics (Fawcett *et al.*, 2012).

2.5. Outsourcing activities: A perspective from SASSA

Government adopts outsourcing to improve public sector productivity through economies of scale and savings (Gurría, 2016). Moreover, outsourcing in the public sector has been used as a government strategy to promote social, environmental and industrial policies (Cane, 2004). Market research at SASSA is used for identifying relevant suppliers and products that best meet the entity's needs. The seriousness of market research depends on risk or the value of requirement. For small outsourcing activities, it is handled by internal researchers through searching previous contractors, and for more complex outsourcing, external researchers are hired to conduct extensive market analysis. Moreover,

SASSA has outsourced different functions with the aim of increasing small business participation in the economy, equity, efficiency and quality of service (SASSA, 2016; Barlow, Roerich and Wright, 2013). SASSA outsources some non-critical activities such as market research, promotion, security, catering, cleaning, logistics, information technology and accounting with the aim of focusing on core competencies and saving organisational resources (SASSA, 2016).

Recently, SASSA and SAPO entered into an outsourcing agreement that costs R2.2 billion per year to provide payment services for all beneficiaries (SAPO, 2017). The service provider responsible is to distribute pension payments in all nine provinces, consolidate one beneficiary database, receive a correct amount, issue social grant cards, and ensure that all beneficiaries prevent fraud and corruption (SAPO, 2017). SASSA and SAPO are in a collaborative effort of promoting a new SASSA card that has no illegal deduction, has free cash withdrawals, is safe to use, and has unlimited pin resets (SASSA, 2017).

2.6. Outsourcing challenges in the public sector

Outsourcing in the South African public sector has been granted constitutional status and is recognised as a means of addressing past discriminatory policies and practices (Bolton, 2006). According to Ambe and Badenhorst-Weiss (2012), reforms in public sector outsourcing were initiated to promote good governance principles and a preferential system to address socio-economic objectives. The reformed process was initiated due a lack of accountability, fragmented processes, and a lack of supporting structures (Ambe and Badenhorst-Weiss, 2012). According to the National Treasury (2005), the fragmentation and deficiencies in terms of governance and the inconsistent interpretation and Preferential Procurement Policy Framework Act 5 of 2000 resulted in the introduction of supply chain management in the public sector as a policy implementation tool. This was the attempt by governments to achieve the desired strategic policy outcomes through outsourcing processes (Arrowsmith, 2010; McCrulden, 2004). Despite the employment of supply chain management and reformed processes as strategic tools, the South African public sector continues to face numerous challenges in its outsourcing practices (Smart Procurement, 2011; Jansen and Stonewash, 2004). Moreover, expenditure in the public sector continues at high levels in the face of ever-increasing concerns about outsourcing practices (De Lange, 2011). There are allegations of corruption and bribery, organisational politics, price inflating, loss of control, risks, hidden costs and quality problems (Smart Procurement, 2011; Jansen and Stonewash, 2004).

2.6.1. Corruption and bribery

Corruption in outsourcing occurs when a tender is given to individuals who are associated with a certain political partner or family members of the top decision-makers in the procurement function (Osei-Tutu

et al., 2010). Increased corruption in the public sector and the mismanagement of tax-payers funds due to poor management in outsourcing processes has cost taxpayers million of rands each year (De Lange, 2011; Shaw, 2010; Mahlaba, 2004). About 20% of government's procurement budget is always missing every year (De Lange, 2011). Munzhedzi (2016) equates corruption and outsourcing in the South African public sector as inseparable twins. Outsourcing has blurred the lines of accountability between public sector and private service providers (Rispel *et al.*, 2015). Government has set up several commissions in both private and public sectors to probe allegations of widespread corruption (National Treasury, 2019). The problem of corruption, negligence and incompetence affects all levels in the organisation. In some parts of the public sector, supply chain management personnel are awarding contracts to their family members, friends or political partners (Osei-Tutu et al., 2010). Like any other public entity, SASSA faces similar challenges. SASSA outsources different activities, ranging from cleaning to security and different business operations (SASSA, 2015).

2.6.2. Organisational politics

The public sector operates in an intense environment, driven by advances in technology, political and public service delivery expectations and programme reviews (Eyaa and Oluka, 2011; Bolton, 2006). Once an outsourcing decision has been made, it is difficult to change that decision due to contract penalties. If organisations decide to in-source functions, they are required to hire well-trained and experienced staff or train existing staff and management so that they are able to control and monitor suppliers. Outsourcing has huge implications for staff morale, such as a loss of core competencies, fear of losing the job, and fear of moving from one company to another due to outsourcing decisions (Van Weele, 2005).

2.6.3. Price hike

Outsourcing was introduced to reduce operating costs, but recent trends in outsourcing indicate that developing countries use it to promote local economic development and opportunities for previously disadvantaged groups (Ambe and Badenhorst-Weiss, 2012; National Treasury, 2005). Moreover, Dhall (2013), claim that outsourcing reduces costs and improves organisational performance. In the process of ensuring economic empowerment, suppliers set prices very high which makes the system vulnerable to not achieving the benefits of cost reduction (Jeppesen, 2014; National Treasury, 2005).

2.6.4. Loss of control

Loss of management control in monitoring and controlling stems from the lack of managing people, processes, power in negotiations and contract management (Gulzhama, 2012). Organisations can lose control due to collusive tendering, complete dependence on external suppliers, and loss of control over the development, preparation and realization of the planned project (Khatik and Khanooja, 2013). To

manage and control suppliers, it is important for both parties to understand deliverables, terms of the contract, compensation and dispute prevention (Du Plessis, 2005). Moreover, it is important to understand the supplier's future goals before signing any contract (Jianna Kleain, Tesch and Chen, 2015).

2.6.5. Risk

According to Hill (2001), risk is defined as the likelihood of a future event occurring. Sullivan and Ngwenyama (2005) define risk as issues which are not addressed appropriately that may lead to negative outcomes. Moreover, effective risk management is something that is important when the organisation decides to outsource in the 21st century (Dinu, 2015; Hood and Miller, 2009). Furthermore, contractors are becoming increasingly risk averse due to slow economic growth. To minimize the risks of poor quality, it is better to avoid a contractor with poor efficiency records (Ambe and Badenhorst-Weiss, 2012). Moreover, sharing risk helps an organisation to shift certain responsibilities to the service provider. According to the National Treasury (2018), outsourcing in the public sector has its own risks, such as a reduction of government accountability and responsibility. Given the increasing economic pressure on South African public entities to reduce spending without reducing the quality of service, the outsourcing process is becoming increasingly risky.

2.6.6. Hidden costs

According to Taylor (2006), hidden costs pose a threat in outsourcing and it generally affects organisations that are heavily involved in outsourcing functions. The organisation and service provider sign a contract that covers the details of the contract, and any costs that are not in the contract are additional costs (Liao and Reategui, 2013). Moreover, reviewing contracts inflicts large financial losses for both sides (Schniederjans *et al.*, 2005; Thoms, 2004). Hidden costs include costs such as the management and coordination of contractors; the cost of an outdated outsourcing strategy; the cost of sub-par inventory performance; the cost of unplanned logistics activities and premium freight; inappropriate sales and operations planning; poor or substandard quality; cost of warranty, returns and allowances; unplanned and unforeseen risks; costs of supplier management; and costs of cash flow (Taylor, 2006; Schniederjans *et al.*, 2005; Thoms, 2004).

2.6.7. Quality problem

Improved quality is one of the reasons for outsourcing, as failure to assure quality can harm the organisation's sales, marketing position and reputation (Ambe and Badenhorst-Weiss, 2012). According to Kruger *et al.* (2007), organisations need to adopt quality control measures, to inspect products or services and compare them with the specifications and set standards, to ensure that the product adheres to the proposed design and specifications. Furthermore, it is very important for

organisations to select the best, most experienced supplier that does not compromise on quality, has an excellent track record, is more flexible in hiring qualified personnel, and rewards employees (Andersson *et al.*, 2006). However, choosing an unreliable service provider may lead to delays, sub-standard quality, deferred delivery time frames and defects in products. Van Slyke (2003) states that outsourced contractors are motivated, better paid, and management standards are generally higher, which leads to organisational efficiency.

2.7. Process and procedures for outsourcing

According to the South African procurement legislative framework, all public entities need to follow the outsourcing processes (Preferential Procurement Policy Framework Act 2000). The steps in public sector outsourcing/procurement include determining the need and specifications; finding, evaluating and selecting the supplier; finalising purchasing contracts; ordering and expediting the order; and lastly, follow-up and evaluation.

2.7.1. Determining the need and specification

Determining the need and specification in the outsourcing process refers to the need for specific goods or services required by the user department (Ambe and Badenhorst-Weiss, 20122). According to Johnson *et al.* (2011), specifications are described as the functionality and technicality of a product, as well as activities that the service provider has to perform. Furthermore, specifications include target budgets, quality specifications, logistics specifications, maintenance specifications, legal and environment specifications (Testa, Annunziata, Iraldo and Frey, 2016). In addition, determining the need and specification has to be through communication with the budgeting section, and the user department needs to plan ahead to avoid placing urgent orders, which often leads to wasteful expenditure (Johnson, Leenders and Glynn, 2011).

2.7.2. Finding, evaluating and selecting suppliers

Finding, evaluating and selecting suppliers is the second step in the outsourcing process aimed at searching for a qualified service provider. This step in the outsourcing process includes tendering, checking of eligibility and bid evaluations (Ambe and Badenhorst-Weiss, 2012).

2.7.2.1. Tendering

According to Weele (2005), a tender is defined as an offer to supply goods or to do work at a fixed price. Tendering is a core component in the outsourcing process, designed to ensure fair procurement practices (Ambe and Badenhorst-Weiss, 2012). According to Johnson *et al.* (2011), there are two types of tendering, namely closed and opened tenders. A closed tender is a tender with a limited number of suppliers and an open tender is a tender without pre-qualification. Moreover, it is the first stage in the

outsourcing process involving an invitation of service providers to bid, requests for proposals and direct purchases from vendors (Johnson et al., 2011). Furthermore, notification on tenders is usually placed in the media, such as newspapers, e-mails, organisational websites and noticeboards, just to mention a few (Roos et al., 2008). In addition, the invitation to service providers to bid involves notifying interested parties to collect the document/s required for application. This includes the place and time for briefing and proposal submissions, closing dates and, lastly, the required documents (Ambe and Badenhorst-Weiss, 2012). Requests for proposals include notifying the most suitable service providers to submit proposals; proposals are evaluated based on organisational efficiencies, quality, capability, reliability, desirable qualities and black economic empowerment (Watermeyer, 2011). The required documents for tendering in the public sector include a valid tax clearance certificate, Broad-Based Black Economic Empowerment (B-BBEE) certificate, company registration documents, organisation's shareholders' certificate, three years' efficiency statements, letter of guarantee from an efficiency institution, proof of public indemnity and, lastly, a letter of good standing with the Department of Labour (National Treasury, 2016). Tendering helps organisations to examine suppliers, maintain competition in the market, comply with procurement policies and due diligence, and ensure better value (Johnson et al., 2011).

2.7.2.2. Bid evaluation

Bid evaluation is a process of opening, identifying and selecting a service provider who will be awarded a contract (Roos *et al.*, 2008). Moreover, bid evaluation of a service provider refers to examination of the authenticity of the documents submitted by services providers, which includes checking references, BEE status, reliability, quality and capabilities, partnerships and the management of the selected supplier (Ambe and Badenhorst-Weiss, 2012). Furthermore, the eligibility of applications is evaluated by a Bid Committee in terms of B-BBEE status, proposals or plans of action, experience and qualifications, and most importantly, suppliers' pricing (Johnson *et al.*, 2011). In addition, the request for proposals enables organisations to compare supplier proposals, as it consists of the rationale of outsourcing, scope, specifications, partners' qualifications required, pricing and expected improvement from suppliers (Roos *et al.*, 2008). This step includes the selection of all the applications with the correct documents and informs unsuccessful suppliers in writing as per the Promotion of Administrative Justice Act (Roos *et al.*, 2008). Selection of a supplier is based on who best meets the requirements, such as timelines, quality, costs and dependability (Ambe and Badenhorst-Weiss, 2012). The company with the highest score in terms of the scorecard is selected and the accounting officer needs to be notified in writing, as do the Provincial Auditor-General and National Treasury (National Treasury, 2016).

2.7.3. The purchasing contract

The purchasing contract is a process whereby two parties enter into an agreement and sign the Service Delivery Agreement (Roos *et al.*, 2008). Prior to negotiation, both parties should be upfront about their strategies and have clear objectives set. In order for negotiations to be effective in the long run, they need to be given enough time to seek win-win results. After appointment of the company, the service provider negotiates the final terms and conditions of the Service Delivery Agreement. If, however, agreement was not reached between two parties, the organisation negotiates with the second best service provider. Managing outsourcing in the public sector requires paying close attention to deliverables, pricing, dispute prevention and terms of contracts (Ambe and Badenhorst-Weiss, 2012). In addition, purchasing contracts focus on measuring performance service level agreements; termination conditions; scope and volume; pricing; resource transitioning; standard performance; factors of production; and control and management (Ambe and Badenhorst-Weiss, 2012).

2.7.4. Ordering and expediting the order

Ordering and expediting are defined as the process of placing an order, and the delivery of goods or services (Edler *et al.*, 2015). This step focuses on awarding the tender to the successful service provider that met all the organisation's requirements (Ross *et al.*, 2008). The order form is considered a legal binding contract. It is important to ensure that information on the order form is correct. Before payment, management and supervisors need to ensure full compliance by contractors regarding conditions and terms of a contract, including criteria of accountability, vendor performance, as well as payments, quality assurance, inspection and other activities that need to be verified (Van Weele, 2009).

2.7.5. Follow-up and evaluation

As the last step in the outsourcing process, this step aims at the monitoring and evaluation of service providers. It is the duty of the buyer to evaluate the supplier's performance, ensuring compliance and recording the user's experience (Roos *et al.*, 2008). This step also ensures that there is a flow of communication throughout the outsourcing process amongst service providers, organisations and other stakeholders (Ambe and Badenhorst-Weiss, 2012).

2.8. Types of outsourcing

An organisation can adopt different forms of outsourcing, which may include the following: business process outsourcing (BPO); information technology outsourcing (ITO); professional outsourcing; manufacturing outsourcing; process-specific outsourcing; business process outsourcing; offshore outsourcing; re-shoring outsourcing; near-shoring; and third-party logistics (Bhagattjee and Hofmeyer, 2009). BPO refers to outsourcing services that do not requires much technical skill, such as marketing, procurement and human resources (Ambe and Badenhorst-Weiss, 2012). ITO refers to outsourcing

functions that require high levels of technical skill, such as the management of infrastructure, organisational systems, application development and the maintenance of organisation systems, amongst others (Bera, 2017). Due to technology advancement and internet connections, outsourcing has developed immensely and has been an excellent solution for cost reduction and faster turnaround time whilst still sustaining quality (Javalgi, *et al.*, 2009). Outsourcing has resulted in the formation of various business alliances with the overall aim of achieving global sourcing and operations.

2.8.1. Professional outsourcing

Professional outsourcing refers to a business strategy whereby a firm hires a specialized company to provide a service (Ray and Ganguli, 2007). According to Bera (2017), professional outsourcing is a business strategy to reduce operating costs, and provide global access to talent while accessing quality service from the supplier. Moreover, professional outsourcing includes specialized services such as information technology, legal, procurement, sales and marketing and others. However, there are some challenges with this business strategy, such as the lack of exclusivity and management difficulties (Javalgi, *et al.*, 2009).

2.8.2. Manufacturing outsourcing

Manufacturing outsourcing is whereby an organisation hires a firm to produce a physical product at a cheaper price than they could have produced it themselves (Bhagattjee and Hofmeyer, 2009). Huge organisations like Nike and Apple are adopting this business strategy by manufacturing their products in China (Fawcett *et al.*, 2012). The benefits of adopting this strategy are cost saving, efficiency and innovation support. However, there are disadvantages, such as ethics and the economic impact in manufacturing outsourcing (Ok, 2011).

2.8.3. Process–specific outsourcing

Process-specific outsourcing refers to renting out a specific internal procedure to the other organisation (Javalgi, *et al.*, 2009). Moreover, it is commonly found in companies such as bakeries to outsource their products to specialized courier companies. According to Bera (2017), process specific outsourcing agreements involve timelines, customer contact details, termination conditions, scope and volume, pricing, resource transitioning and standards of performance. Furthermore, this type of business strategy helps the organisation to focus on their core competency and improve customer service, while reducing time and costs (Andone *et al.*, 2010).

2.8.4. Business Process Outsourcing (BPO)

Business Process Outsourcing refers to a strategy of hiring external suppliers to provide a specific operational activity (Andone *et al.*, 2010). It is a common practice in manufacturing and includes

cleaning services, facilities management and landscaping. Organisations adopt this business strategy in order to focus on their core competency. Moreover, organisations may adopt BOP to gain access to highly skilled personnel and reduce operational costs (Javalgi, *et al.*, 2009).

2.8.5. Offshoring

Offshoring occurs when a company performs a production or service function in another country. In the beginning, offshoring motives were for cost savings, but now have become a widespread competitive strategy (Ok, 2011). The organisation needs to choose a supplier that possesses greater efficiency in order to achieve cost savings due to economies of scale. There is a growth in the number of companies that practice global outsourcing. Hence, the importance of environmental scanning of the global market is crucial. Competition in the economy is intensive and companies are forced to work even harder to maximize productivity and profitability (Javalgi, *et al.*, 2009). Furthermore, international outsourcing is often confused with offshoring. It refers to the relocation of goods manufacturing abroad and the organisation decides if they want control or a joint venture (Javalgi, *et al.*, 2009). The international outsourcing companies do not share profits (Belcourt, 2006). Therefore, organisations outsource goods or services from different countries, sometimes opening international offices and hiring local employees to work. In addition, the cost benefit does not realize immediately in offshoring (Martorelli, 2007).

2.8.6. Re-shoring outsourcing

Re-shoring refers to a process of transferring business operations back to its home country (Martorelli, 2007). In the recent past, organisations were outsourcing their production services globally to achieve reduced labor costs, tap into new markets and access unavailable talent locally. Currently, organisations are realizing that offshoring is not as cost-effective as it used to be due to political, socio-cultural, technological, economic and environmental factors. Hence, organisations are insourcing their production services back due to external regulations that prohibit certain activities within that country and the impact of shipping costs due to rising costs of oil (Martorelli, 2007).

2.8.7. Nearshoring outsourcing

Nearshoring refers to the relocation of business processes in a more profitable area (Javalgi, *et al.*, 2009). Businesses usually relocate business processes to the nearest countries that share similar cultures and geography (Andone *et al.*, 2010). For example, Apple companies outsource manufacturing processes to China to access highly skilled labour and lower costs in production (Andone *et al.*, 2010). According to Andone *et al.* (2010), the benefits of this business strategy are cost savings in terms of labour costs, tax and shipping costs, amongst others.

2.8.8. Third-party logistics (3PL) outsourcing

Third-Party Logistics outsourcing refers to the idea of combining different logistics activities such as distribution, inventory management, warehousing, customer services and procurement (Fawcett *et al.*, 2012). Logistics management helps organisations to access the right product at the right time and right place. Moreover, 3PL enables organisations to anticipate changes, assess and manage risk (Ambe and Badenhorst-Weiss, 2012). The benefits of adopting 3PL are specialization access, spreading of risks, core competency focus, gaining of competitive advantages and direct selling (Ambe and Badenhorst-Weiss, 2012).

2.9. The dimensions of outsourcing

The dimensions of outsourcing are viewed from a business perspective. Some organisations may consider outsourcing the staff or employees, the company being outsourced to, and the communication technology used, as fundamental dimensions. Other organisations may outsource to gain knowledge from a service provider, culture, government and economy.

2.9.1. The staff of outsourcing/outsourced organisations

Outsourcing is not loyal to employees. Usually, the staff of the outsourcing organisation are transferred to the outsourced organisation. Furthermore, outsourcing is used to reduce salary costs, staff employment and eliminate the rights and conditions acquired by employees through union negotiations. One of the major risks of staff outsourcing is the effect on employee morale and performance (Andone, Pavaloaia and Cuza, 2010). For example, human resource outsourcing usually involves restructuring and always displaces employees. Moreover, a reduction of quality and accountability is one of the weaknesses of staff outsourcing. Organisations give employees a sense of recognition and feeling of security and belonging. When these assurances are disrupted, employees feel bitter. Professionals oppose outsourcing because of the threat of losing jobs, and the worry that management believes outsiders to be more capable (Andone *et al.*, 2010).

2.9.2. The company being outsourced to

The company being outsourced to inherits new employees; this allows the organisation to achieve its desired objectives whilst lowering costs since the new employees do not require much training (Andone *et al.*, 2010). Moreover, it allows the new owner to choose the best candidates and train the new staff using the strategies desired, rather than relying on the previous strategy. At the same time, the disadvantages of inheriting new staff are providing accommodation, extra office equipment, space, and other office supplies. It also involves paying some part of the benefits of new employees such as medical aid, taxes, retirement plans and other social contributions (Andone *et al.*, 2010).

2.9.3. The communication technology used

Communication is defined as an act of exchanging ideas, and giving and receiving opinions and information so that the message can be clearly understood by both parties (Lamb, Hair and McDaniel, 2012). It involves the use of letters, faxes, emails, reports, phone calls, memos and meetings, amongst others. Communication is considered amongst the factors that create challenges in outsourcing. Therefore, both parties communicate frequently (Outsourcing centre, 2014). Outsourcing activities should be widely communicated to all potential suppliers through various means of communication (Orville, Walker and Mullins, 2012). New telecommunication, internet and networking technologies provide organisations with an ability to measure everything, optimize spending, syndicate information, and benefit from global reach 24 hours a day, instant communication, and the delivery of products or services (Orville *et al.*, 2012). Using all types of media, SASSA advertises procurement plans such as information on specifications for a standard product, in order to provide potential suppliers with all the necessary information.

2.9.4. The knowledge of a service provider

Knowledge is one of the vital factors in make or buy decisions, even if the organisation is getting lower labour cost benefits from outsourcing. Without proper knowledge or information, organisations need to in-source. Suppliers without experience and knowledge waste organisational resources. It is crucial for procurement and supply chains to empower themselves with knowledge and expertise in order to get value for money from service providers (Forrer *et al.*, 2010). There is no need for organisations to continue outsourcing their functions because it will be a waste of resources. Organisations with the necessary skills and experience should insource their function, because one of the reasons for outsourcing is to search for greater experience and knowledge (Qu and Brocklehurst, 2003).

2.9.5. Culture

Culture is defined as set of values, beliefs and behaviours commonly held by members of one organisation (Abu-Jarad, Younis, Yusof and Nikbin, 2010). Culture enhances organisational performance by the motivation of employees guiding other employee's behaviour. According to Abu-Jarad, *et al.* (2010), organisational performance is defined as an organisation's ability to achieve its goals using less resources efficiently. In addition, culture plays an important role in organisations' outsourcing; it includes organisational norms, values, religion, daily habits and beliefs of the suppliers (Hofstede, Hofstede and Minkov, 2010). However, it is important for both organisations and service providers to understand each other's cultural differences when entering into an outsourcing agreement (Winkler, Dibbern and Heinzl, 2008).

2.9.6. Government

The political stability of a nation determines, to a certain degree, the investment attraction of companies. Countries that are perceived to be politically stable tend to improve their economic and market share. Developing countries offer tremendous opportunities. For example, the Chinese economy is the fastest growing economy in the world, with a 25.1 trillion dollar GDP, followed by the United States of America with 20.2 trillion dollars, while India and Russia are in the top five with 5.5 and 4 trillion dollars' purchasing power parity respectively (International Monetary Fund, 2018). Government ensures political stability in a country in order to attract stable investments (Van Der Linden and Hengveld, 2008). In addition, laws and regulations from other countries can complicate outsourcing processes. For example, China has political and social unrest that increases manufacturing costs and disrupts manufacturing processes in the country (Burgman and Friedle, 2016). There have been calls to restrict Chinese and Brazilian imports in South Africa in an effort to boost South African manufacturing industries (Jurevicius, 2016). Furthermore, pollution and other environmental issues are facing outsourcing, especially in the manufacturing industry. Therefore, it is important to acquire local knowledge before entering any global business (Ambe and Badenhorst-Weiss, 2012; Alon, Herbent and Munoz, 2007).

2.9.7. Economy

Economy is defined as the marketplace that deals with flows of money, including trade, distribution, goods consumption and production of a state, organisation and individuals (Jurevicious, 2016). Moreover, organisations have to pay close attention to the prominent factors driving the economy of the country, such as natural resources, human resources, technological development, capital formation, political and social factors (Jurevicious, 2016). World global outsourcing is worth more than 88.9 million dollars and the larger portion of that outsourcing comes from emerging countries such China and India (Statista, 2017). Different countries around the world are using outsourcing to increase efficiency and reduce operating costs (Ukokobili, 2010). Market conditions or the economy have a great influence in outsourcing, as these determines whether the accomplishment of socio-economic objectives is achieved or not. In the South African public sector, outsourcing is a key mechanism and tool addressing transformation, environmental aims and socio-economic developments (Turley and Perera, 2014).

2. 10. Classification of outsourcing

For an organisation to benefit from outsourcing, it needs to classify outsourcing and choose an appropriate outsourcing strategy that can yield positive results. According to De Villiers *et al.* (2012), organisations can select from different outsourcing strategies such as tactical, strategic, traditional and solution outsourcing.

2.10.1. Tactical outsourcing

Tactical outsourcing refers to the hiring of a firm to perform a specific activity (Khan, 2014). Tactical outsourcing allows an organisation to keep control and oversight of the project or an activity (Galici, 2016). This type of outsourcing strategy is ideal for short-term projects, commonly for short-term manufacturing business. Moreover, technical outsourcing is usually a reactive or routine approach to purchasing and it skips the long outsourcing process of searching for a qualified supplier. The organisation is more knowledgeable of the project and maintains the planning process, such as gathering of requirements and designing a system for a supplier (Cohn, 2015).

2.10.2. Strategic outsourcing

Strategic outsourcing refers to business collaboration/partnerships between contractors and suppliers (Cohn, 2015). Strategic outsourcing is guided by three key components, namely; supplier selection/evaluation, market research and relationship management. Unlike tactical outsourcing, strategic outsourcing is ideal for a long-term project and it requires the organisation to search properly for a qualified supplier (Khan, 2014). According to Galici (2016), the benefit of strategic outsourcing is flexibility, operational cost reduction, and improved effectiveness and efficiency in business operations. Similarly, the disadvantages in strategic outsourcing are a lack of motivation from employees, decreased knowledge of the process, and a lack of control of the final product (Galici, 2016).

2.10.3. Traditional outsourcing

Traditional outsourcing refers to hiring service providers to provide basic activities (Plotkin, 2018). This type of outsourcing helps organisations to focus on performance improvement and the reduction of costs (Fawcett *et al.*, 2012). The focus has shifted from traditional outsourcing to automation and transformation.

2.10.4. Solution outsourcing

Solution outsourcing refers to partnering with a well-established service provider (Cohn, 2015). This type of outsourcing strategy allows a business to improve production standards, quality and efficiency (Ambe and Badenhorst-Weiss, 2012). Solution outsourcing allows the organisation to join forces with a service provider that has already invested in significant research and development. Furthermore, solution outsourcing can lower costs, improve production and is ideal for manufacturing projects (Axenic, 2016).

2.11. Conclusion

In recent decades, the concept of outsourcing has emerged as a major trend in organisations all over the world. Increasingly, more and more companies are focusing on outsourcing to ensure a competitive edge in their core businesses and improved economic development. This chapter provided an introduction, explored existing literature and provided in-depth knowledge on the area of outsourcing. The information provided covers a historical perspective on outsourcing, theoretical models underpinning the concept of outsourcing, namely Relational View Theory and Core Competency Theory, emerging trends, and outsourcing activities from a public sector perspective. In addition, this chapter addressed outsourcing activities narrowed to the case study organisation, and challenges facing outsourcing in the public sector. Further, this chapter discussed processes and procedures for outsourcing, and provided information on the types, dimensions and classification of outsourcing. The next chapter is research methodology and provides information on how the research was conducted.

Chapter Three

Research Methodology

3.1. Introduction

The previous chapter focused on a literature review of the concept of outsourcing. This chapter presents the way in which the study was conducted, commonly referred to as the research methodology. According to Sekaran and Bougie (2016), research methodology is defined as an overall strategy that integrates different components of the study in a logical way. The research methodology is important in conducting a study because it ensures that the research problem is effectively addressed. Given the information on research methodology, this section provides the introduction to the chapter, research design and paradigms, research methodology, and study sampling. The trustworthiness of the study is discussed in terms of reliability and validity, and lastly the conclusion is provided.

3.2. Research design and paradigms

Research design has been defined by scholars, from different perspectives. Vosloo (2014) defines research design as a plan or blueprint for the collection, measurement, and analysis of the data of the study, its nature, validity and reliability, which helps a researcher to answer research questions. In the view of Wiid and Diggines (2010), research design is simply the framework, plan or outline. From the above two definitions, research design deals with the gathering and analysis of data to answer the questions and objectives of the study. The term design in this context refers to the environment, such as people, places, activities or objects that are investigated (Fink, 2002). The aim of research design is to ensure that the data collected allows the researcher to efficiently address the research problem in a logical and critical manner (De Vaus, 2006). According to Wiid and Diggines (2010), research design provides the researcher with information about the sample to be used, groups and the necessary variables. The primary objective of the research design is to make sure that the problem is researchable by designing the study in a manner that will provide answers to specific questions.

The concept of the research paradigm is defined as a system that involves organised thinking and practices to define the nature of enquiry using methodology, epistemology and ontology dimensions (Terreblanche and Durrhiem, 1999). Research paradigms are characterized through methodology, epistemology and ontology (Guba, 1990). There are various types of research designs, namely exploratory, experimental, methodological, historical and descriptive. As an exploratory study, this study adopted an exploratory research design. The researcher's reason for adopting the exploratory research design stemmed from the fact that not much is known about the situation at hand, or no information is available on how similar problems or research issues have been solved in the past. Exploratory research is flexible in nature. As a result, findings cannot be generalised (Sekaran and

Bougie, 2016). According to Creswell (2014), exploratory research often relies on a qualitative approach of data gathering, such as case studies, focus groups, interviews and informal discussions. The exploratory study also helps in gaining the insight needed about the phenomenon, problem or opportunity.

3.3. Research methodology

A research methodology is an analytical way of solving a research problem, a tool used to gain knowledge (Creswell, 2014; Denzin, 2010). From another school of thought, research methodology provides a plan on how to conduct the research effectively and techniques to be used for the collection of data (Silverman, 2011). It explores various themes such as data collection instruments and methods (Cohen *et al.*, 2013). There are four main approaches to a research study, namely quantitative research, advocacy, mixed approach and qualitative research (Sekaran and Bougie, 2016). Quantitative research is defined as a process of collecting large amounts of data, with more representative respondent samples and the numerical calculation of results (Cohen *et al.*, 2013; Monette, Sullivan and Dejong, 2011). According to Sekaran and Bougie (2016), the advocacy research approach is defined as a process of collecting data with the intention of providing argument or evidence to support a particular cause. A mixed approach is defined as a process of collecting data using both qualitative and quantitative research methods (Creswell, 2014; Monette *et al.*, 2011).

This research study adopted a qualitative approach. In line with Creswell (2014), the qualitative method is suitable for this study as it offers an in-depth understanding and rich information of the phenomenon. This method utilises all question types and generates rich detailed data, whether it is gathered from structured, unstructured and semi-structured interviews (Creswell, 2014). According to Silverman (2011), the qualitative research method is defined as a process of collection, analysis and interpretation of data that cannot be meaningfully quantified and summarised a data in the form of numbers. Sergi and Hallin (2011) state that qualitative research is used for non-numerical data, is less structured and therefore uses smaller sample sizes. Qualitative research relies on detailed descriptions by respondents to gain insight into a problem or phenomenon, explores issues, and grows understanding of the reason behind the issues (Sekaran and Bougie, 2016; Shannon, 2005). As such, qualitative research helps the researcher to explain and describe the current situation and therefore can be considered confirmable, dependable, transferrable and credible (Hair et al., 2013). Qualitative research collects data using instruments such as material culture, analysis of documents, observations, in-depth interviews and focus groups (Bernard and Ryan, 2010). Moreover, qualitative methods allow a researcher to investigate the motives and the rationale for outsourcing functions in the public sector. This research study adopted a qualitative approach aimed at understanding the phenomena, exploring issues and answering the research questions. The main advantage of adopting qualitative research methodology is that it enables a researcher to have a clear understanding of the research problem (Sekaran and Bougie, 2016; Silverman, 2006; Tucker, Powell and Meyer, 1995). Lastly, qualitative research is well suited for this study because it permitted the researcher to identify the barriers related to outsourcing functions in the South African public sector at the South African Social Security Agency, Umzumbe local office.

3.4. Study Sampling

According to Martin (2005), sampling is defined as a process of selecting participants from a total population in a way that represents the whole population. Yin (2012) defines a sample size as the overall number of participants who take part in the study or the sample size that represents the population under study. Sampling processes usually include the population and sample of the study, sampling strategies, data analysis and reliability and validity of the study. The sampling frame includes the list of elements from which the sample is actually drawn (Yin, 2012; Cooper and Schindler, 2006). The following are addressed under this section: population, targeted population and sample size.

3.4.1. Population

Choosing the population is an important step in the research process, which allows the researcher to select an appropriate and representative sample (Monette *et al.*, 2011). According to Wiid and Diggines (2010), a population is a group of people from whom the data is needed. It may consist of a number of individuals, or items that can become objects for observation (Hair *et al.*, 2013). The population for this study are 300 employees from Umzumbe local office.

3.4.2. Targeted population

A target population involves a specified group of people or objects for which questions can be asked or observations made to develop required data structures and information (Monette *et al.*, 2011). The target population for this study are the 90 employees from all the five units at SASSA Umzumbe local office. The researcher considered only permanent employees who have been with the organisation for two or more years because of experience and in-depth knowledge of SASSA's policies, processes, and procedures relating to outsourcing.

3.4.3. Sample size

A sample size is the total number of people or units or selected to participate in the study (Yin, 2009). In this study, the sample size is comprised of 41 permanent employees selected from management and non-management levels in the department. The sample were selected from the following departments: Supply Chain Management, Finance, Human Resources Management, Information and Technology and Marketing. The researcher conducted a face-to-face interview with six well experienced managers from

the five identified units and focus group discussions with thirty-five junior staff at SASSA's Umzumbe Local Office.

3.4.4. Sampling strategies

According to Simons (2010), sampling strategies are defined as methods of selecting a sample within a general population with the aim of eliminating bias in the findings of the study. There are two types of sampling strategies, non-probability sampling and probability sampling (Hair *et al.*, 2013). In non-probability sampling, people and units are selected based on the researcher's judgement (Thomas and Brubaker, 2000). Yin (2012) states that a probability sampling strategy gives the researcher a chance to create a sample that truly represents the population. According to Cooper and Schindler (2006), non-probability sampling consists of three different methods, namely, purposive sampling, quota sampling and convenience sampling. Probability sampling consists of simple random sampling, cluster random sampling, stratified random sampling and systematic sampling (Creswell, 2014). For the purpose of this study, the researcher adopted the qualitative approach and a non-probability purposive sampling method. Purposive sampling concentrates on particular people within a population to be used for a study, which will provide relevant information on the study (Enkvillage, 2015). The purposive sampling method is considered the best method when the researcher is looking for a certain characteristic in the population (Business Analytics, 2015). The method is considered ideal for the study because only a limited number of people can serve as primary data sources.

The researcher chose a purposive sampling method to be able to gather data from respondents. The researcher believed that the selection of the respondents across the management level would provide in-depth knowledge into the research questions.

3.4.5. Data collection methods

For this study, primary qualitative data were obtained by means of focus group discussions and face-to-face interviews with employees at the SASSA Umzumbe local office. Interviews and focus group discussions allow the researcher to investigate issues in an in-depth way and follow-up on incomplete answers (Monette et al., 2011). The semi-structured interviews consisted of several key questions that helped to define the areas to be explored, but also allowed the researcher to use probing questions. Through the use of focus groups discussions and face-to-face interviews, the qualitative method assisted the researcher to capture a great deal of information and compare responses. Exact verbal accounts of the respondents' experiences and perceptions were recorded. Interviews and discussions were audio-recorded then transcribed and also the interviews held in English and Zulu, and then translated and transcribed. The researcher had direct contact with participants, hence, it was easy to understand the phenomenon under study.

According to Creswell (2014), a face-to-face interview refers to a data collection method that directly communicates with people personally as well as telephonically. Similarly, a focus group discussion refers to a process of collecting data on a particular concept or topic using group discussions (Monette *et al.*, 2011). The researcher's reasons for adopting face-to-face interviews and focus group discussions are as follows:

- A face-to-face interview allows for direct interaction with the respondents and clarifies, rephrases and adapts the questions as necessary;
- Focus group discussions provide an opportunity for free flowing and flexible discussions, and;
- Focus group discussions allow spontaneous and unstructured responses from respondents.

The research instrument adopted for this study is an interview schedule. The interview guide questions used for the study were comprised of three sections.

Objective one aimed to answer the research questions addressing the barriers facing outsourcing functions in the South African public sector. The researcher asked the respondents questions one, four, and six (See Appendix A) during the collection of data. These questions were formulated with the aim of identifying the barriers facing outsourcing functions in the public sector.

Objective two aimed to answer research questions addressing the extent to which outsourcing-related challenges impact on the organisation's operations. The researcher asked respondents questions three, five, eight and nine during data collection. These questions were formulated with the aim of determining the extent to which outsourcing-related challenges impact on the organisation's operations.

Lastly, Objective three aimed to answer the research question addressing the existing monitoring measures and their impact on the outsourcing functions in the South African public sector. The researcher asked the respondents questions two, seven, ten, eleven and twelve during the collection of data. These questions were formulated with the aim of understanding the existing monitoring measures and their impact on the outsourcing functions in the South African public sector.

3.4.6. Overview of data analysis

According to Sekaran and Bougie (2016), data analysis is defined as a process of unlocking the information hidden in raw collected data and transferring it into something meaningful and useful. The objective of data analysis is to discover meaning, themes, concepts and patterns (Yin, 2012). Data analysis has multiple facets and approaches, encompassing diverse techniques under a variety of names, in different business, science and social science domains (Yin, 2012). A data analysis process helps the researcher to summarize, organize and make sense of collected data. In the data analysis process, the

researcher identifies themes, patterns and explanations, and interprets these with the aim of making sense of the collected data. Furthermore, in data analysis, the researcher is inspecting, cleaning, transforming and modelling data with the goal of highlighting useful information, suggesting conclusions and supporting decision-making (Bogdan and Biklen, 2003). According to Creswell (2014), it is important to check the quality of data collected and arrange data in a manner that allows smooth analysis. In the present study, questions were discussed, and notes were cross-checked for completeness and accuracy.

This study adopted content analysis to analyse the raw data obtained from the focus group discussions and face-to-face interview sessions with the respondents. Content analysis is a systematic, rigorous approach that analyses data obtained and generates meaningful information (Monette et al., 2011). According to Sprague (2005), content analysis involves classifying and coding data with an aim to make sense of the data collected. Content analysis enabled the collected information from both interviews and focus group discussion to be treated in a way that made it possible to interpret the barriers facing outsourcing activities in the public sector. Raw data obtained in data collection is converted into information that helps a researcher to answer a research question and make useful decisions (Monette et al., 2011). To obtain meaningful information, data emerging from a study were coded, analyzed and re-arranged to form themes, clusters, and categories. Content analysis is a highly flexible research method that has been widely used in different studies to achieve research goals and objectives (Monette et al., 2011). According to Monette et al. (2011), content analysis can be employed with a wide range of analytic techniques to generate findings and put them into context. There are two general types of content analysis; relational and conceptual (Monette et al., 2011). The study adopted a relational analysis, where the researcher identified research questions and chose a concept, and explored the relationships between the concepts.

3.5. Reliability and Validity of the study

According to Sekaran and Bougie (2016), reliability and validity concepts are the most important step in a research process, and are applied by researchers to assess the quality of a study. Reliability is a characteristic of measurement concerned with accuracy, precision and consistency (Denzin and Lincoln, 2017). Reliability ensures a sense of standardisation and deals with the dependability, stability and consistency of the research instruments (Monette *et al.*, 2011; O'Leary (2004). Due to the fact that research studies face challenges of respondent biases, reliability addresses this challenge by applying a particular technique repeatedly to the same object in order to achieve a consistent unbiased results (Sekaran and Bougie, 2016; Neuman, 2012). Validity is defined as an instrument that measures truth and accuracy, as well as the truth and accuracy of the results drawn from the data (Sekaran and Bougie, 2016). According to Creswell (2014), validity concludes whether the research accurately measures that

which it was planned to measure or how truthful the research results are. Nueman (2012) defined validity as the extent to which conclusions drawn from the research provide an accurate explanation and reason for what happened.

The types of reliability commonly used in research studies are inter-rater, test re-test and inter-method. The study addressed inter-method reliability, which refers to assessment of the degree to which test scores are consistent when there is a variation in the instrument or methods used (Denzin and Lincoln, 2017). Specifically, the researcher aimed for homogenous inter-method reliability. In this case, the researcher used the same questions for the face-to-face interviews and focus group discussions. The researcher chose this to stem shortfalls from the responses in either of the approaches. It also allowed for comparison of the views of the management and the non-management respondents.

In measuring the reliability of the research instrument, the respondents did not have issues answering the questions, thus reducing any bias or ambiguity. Furthermore, the supervisor reviewed the research instrument prior to data collection, to ensure proper alignment of the questions with the research objectives.

Validity is premised on the assumption that what is being studied can be measured; it confirms the accuracy and the truth of the measured and captured data and findings drawn from the data (Nueman, 2012). The types of validity commonly used in research studies are content validity, ecological validity, internal validity, construct validity and external validity (Sekaran and Bougie, 2016). Within the limits of this study, the researcher adopted content validity. Content validity refers to a replicable and systematic technique of compressing information into minimal content that fully covers the objectives of the study (Sekaran and Bougie, 2016). Straub *et al.* (2004) define content validity as the degree to which items in an instrument reflect the content universe to which the instrument will be generalized. Nueman (2012) further stated that the assessment of the overall validity of conclusions drawn from a study can be made by providing the following three aspects: validity of explanation, validity of measurement and validity for generalisation.

- Validity of explanation/internal validity asks whether the contexts and conclusions derived from the research are correct.
- Validity of measurement asks whether the research instrument measures what it supposed to
- Validity of generalisation/external validity asks whether the conclusions drawn from a particular study can be generalised to other contexts or people.

In this study, the choice of a research design using triangulation of data, and the pre-testing and use of the questionnaire and interview schedules were found suitable to ensure reliability and validity of the research findings and conclusions made. Errors on the questionnaire were corrected and some questions were rephrased after pre-testing to make them easier for the respondents to understand. Furthermore, the secondary data were sourced for reputable journal articles, government publications, internet sources and textbooks.

3.6. Conclusion

This chapter has discussed the method used in gathering the information for the study. The chapter covered the research design and paradigms, research methodology, and study sampling. It discussed the reliability and validity of the study and the conclusion. The next chapter discusses the research findings of the study, based on the research objectives.

Chapter Four

Presentation and Analysis of Data

4.1. Introduction

The previous chapter presented the methods by which the research was conducted, explained the research design and highlighted the research approach used for the study. Moreover, it explained the data collection techniques adopted. Furthermore, this chapter includes the data presentation and analysis, response rate, test of the research objectives and a discussion of the research findings. The main aim of this chapter is to summarise the primary data and to interpret the data that were collected using face-to-face interviews and focus group discussions. This chapter presents an analysis of the findings. This study focused on the outsourcing of functions in the public sector. The study is guided by three research objectives, namely:

- To investigate the barriers facing outsourcing functions in the South African public sector;
- To determine the extent to which outsourcing related challenges impact on the organisation's operations, and;
- To examine the existing monitoring measures and their impact on the outsourcing functions in the South African public sector.

4.2. Presentation and data analysis

According to Sekaran and Bougie (2016), data analysis is a process of minimizing collected data to a manageable size. As mentioned earlier in chapter four, section 4.3.1, the research approach adopted for the study is qualitative in nature and content analysis was used in the interpretation of the data collected. Content analysis is defined as a method used to interpret text content through the systematic processes of classification, coding and the identification of patterns (Hsieh and Shonnon, 2005). According to Neuman (2012), classification refers to a process of categorising data according to types or groups. Identification of patterns refers to the identification of traits or recurring characteristics which help the identification of problems (Creswell, 2014). Sekeran and Bougie (2016) define coding as a process of analysis, re-arranging and reducing collected data from the semi-structured in-depth interviews and focus group discussions. The researcher used themes to examine and record patterns in the research. Themes are defined as patterns that emerge from examining, pinpointing and recording patterns in research (Creswell, 2014) (See Table 4.1). Research respondents were grouped under the following departmental categories; Supply Chain Management, Finance, Operations, Human Resource Management, Information Technology and Marketing

TABLE 4.1: Demographic of respondents of the Face-to-Face Interview session

Respondents	Interview	Position	Units/Section	Gender	Years of	No.	No.
	schedule				experienc	Targeted	sampled
					e		
R1	Face-to- face	Senior	Operations	Male	16 years	3	1
	interview	Management					
R2	Face-to- face	Senior	Supply Chain	Female	10 years	3	1
	interview	Management	Management				
R3	Face-to- face	Senior	Marketing	Female	8 years	3	1
	interview	Management					
R4	Face-to- face	Senior	Finance	Male	12 years	2	1
	interview	Management					
R5	Face-to- face	Senior	Human	Male	10 years	3	1
	interview	Management	Resources				
			Management				
R6	Face-to- face	Senior	Information	Male	9 years	2	1
	interview	Management	Technology				
Total		•	•	<u> </u>	•	16	6

Source: Compiled by the researcher (2019)

As shown in Table 4.1, the study comprised six respondents who are working at SASSA. Amongst the six respondents in the study, the total number of females is two and four are males. Moreover, all the respondents have more than 2 years' working experience.

TABLE 4.2: Demographic of respondents of the Focus Group discussions

Respondents	Interview	Position	Units/Section	Gender	Years of	No.	No.
	Schedule				experience	targeted	sampled
R1	Focus	Administrators	Supply Chain	8 males and 4	All above 2	35	12
	group	and	Management	females	years		
	discussion	Supervisors					
R2	Focus	Administrators	Marketing	5 males and 6	All above 2	28	11
	group	and		females	years		
	discussion	supervisors					
R3	Focus	Administrators	Finance	7 males and 5	All above 2	27	12
	group	and		females	years		
	discussion	Supervisors					
	<u>'</u>	1	<u>'</u>	1	1	90	35

As shown in Table 4.2, the study comprised thirty five respondents who are working at SASSA. Amongst the thirty five respondents in the study, the total number of females is fifteen and twenty are males. Moreover, all the respondent have more than 2 years' working experience.

4.3. Response Rate

Response rate is defined as the number of people who participated in a study (Neuman, 2012). According to Bryman (2008), a response rate of 70% is considered very good for analysis, 60% is considered good, while 50% is considered enough. According to Creswell (2014), lower response rates are likely to have a greater risk of bias in the findings. For face-to-face interviews, sixteen senior management were targeted, and six management participated. Therefore, there was a 38% response rate. For focus group discussions, a total of ninety junior staff were targeted and thirty-five participated in the study, a 39% response rate. The overall response rate is therefore 77%.

4.4. Test of the Research Objectives

The researcher used the same questions for both the face-to-face (FTF) interviews and focus group discussions (FGD) (See Appendix A). The common findings from both approaches were presented together as per each research objective.

4.4.1. Test of Research Objective One:

Research objective one was structured to ascertain the barriers facing outsourcing in the public sector. The questions that provided answers for this research objective are question one, four and six (See Appendix A). These questions were structured to ascertain the on-going challenges facing outsourcing in the South African public sector. The questions thus ask:

Question 1: Do you think the number of staff working in the supply chain management is sufficient given the size and needs of the organisation?

Data emanating from the FTF interview indicated human resource incapacity and a failure to replace and hire new staff. As one of the respondent noted (R7):

"No, there is no proper planning in a procurement unit and the unit is understaffed and if one of the staff members retires or leaves a job, it is not easy to replace them due to budget cuts regulations. Moreover certain procurement officers are taking strain by performing certain tasks which are not in their job description.

Further, FGD revealed concerns with the number of staff in the supply chain management unit. As one of the discussant noted (R8):

"Yes, the staff at the supply chain management section is not sufficient and there is a lack of capacity within the human resource department when hiring supply chain officers. This challenge leads to poor job execution resulting in a backlog in the procurement process."

As the discussion proceeded, the researcher asked a further question to ascertain the supply chain management unit educational capacity (See interview schedule, question four).

Question four: Does management have proper qualifications and training in in relation to outsourcing?

Data emanated from FGDs indicated that there is shortage of skills, knowledge and competencies in a procurement section. As put by one of the discussant (R6):

"Management in the procurement unit does not have proper qualifications and training in the supply chain management, which usually leads to selection of inefficient suppliers and appointment of incompetent individuals in the top positions."

Further, in the FTF interviews, data indicated concerns with the lack of proper supply chain management qualifications, skills and training, as noted by one of the discussants in a FTF (R7):

"There is a lack of skills in any of the top positions, but we have experience and other management related qualifications."

The researcher probed with a question aimed at evaluating the benefits and remuneration of government officials (see interview schedule, question six). The questions asks:

Question six: How are managers and staff rewarded for achieving supply chain management objectives and improving the value delivered to clients?

Data emanating from FGDs indicate concerns with salaries of government officials. As one of the discussants noted (R5):

"We are not getting paid enough compared to the private sector, that is why some of the government officials are leaving our department and they are not replaced due cost cutting measures, which ultimately affect the standard of service we provide to clients".

Data emanating from FTF interviews indicate that staff rewards in the public sector do not motivate. As one of the respondent noted (R7):

"We are not happy with salaries and benefits in public sector, the reason we are still working here is because of job security."

Emerging from the data are concerns around lack of proper planning, incompetency in the SCM unit, lack of qualifications, employees not being well paid compared to the private sector, and human resource incapacity. Thus the researcher grouped these under the "Theme Cluster 1: Barriers."

4.4.2. Test of Research Objective Two:

Research objective two was structured to ascertain the extent to which outsourcing-related challenges impact on the organisation's operations. Questions three, five, eight and nine provided answers for this research objective (See Appendix A). These questions were structured to ascertain the extent of outsourcing-related challenges and their impact in the South African public sector.

Question three: Are the departments outsourcing of functions ensuring organisational continuity; if not, why does the department outsource?

The data from FTF interviews with one senior manager indicated that the outsourcing of activities has a negative implication on the organisation's continuity. As put by one of the respondent (R6):

"No, outsourcing of activities does not yield the benefits and there are negative implications on the organisational continuity due to unethical practices, poor procurement management practices and

service provider control which affect the quality of service provided by suppliers. Moreover, department outsources with an aim of addressing socio-economic objectives."

The FGD data reveals concerns with rotations of suppliers and BBBEE validation criteria that ensures suppliers are able to meet the organisation objectives and achieve the socio-economic objectives with black-based suppliers from previously disadvantaged backgrounds. As one of the discussant noted:

"No, outsourcing of activities does not have positive implications in an organisation. The reason for outsourcing is to address the socio-economic challenges through selection of suppliers from previously disadvantage backgrounds."

The researcher probed a question aimed at evaluating the organisation's adoption of new technologies to improve outsourcing processes (see interview schedule, question five). The questions thus ask:

Question 5: Does the organisation readily adopt new technologies to improve on the outsourcing processes?

The data emanating from FGDs indicated that the organisation does not readily adopt new technologies to improve on the outsourcing processes. As one of the discussants noted (R9):

"The department is using Standard Chart of Account to capture purchase orders and invoices on Basic Accounting Systems; there is no alternative systems and if the system is down there is nothing an official can do."

The data emanating from FGDs indicated that the organisation readily adopts new technologies to improve on the outsourcing processes. As one of the respondent noted (R5):

"All the government departments are using Basic Account Systems and the system is updated frequently to ensure that it improves the procurement processes."

The researcher probed a question aimed at evaluating the impact of outsourcing functions (see interview schedule, question eight). The questions ask:

Question eight: How would you rate the impact of outsourcing to the organisational goals and objectives?

Data emanating from FTF interviews revealed that outsourcing of activities does not have a positive impact on the goals and organisation objectives. As one respondent noted (R1):

"The department outsourcing of activities does not have a positive impact on the organisation objectives and goals because inefficiency, lack of access to highly skilled service providers; and cost cutting advantages associated with outsourcing is not achieved due to price inflation, corruption and fraud in a procurement processes."

Similarly, FGDs also voice concerns about communication with various units and planning. As one respondent noted (R6):

"Outsourcing has a negative impact on the goals and objectives of the organisation due to poor planning and miscommunication between various sections."

The researcher probed with a question aimed at evaluating the organisation's effectiveness in communicating the value of outsourcing to all the stakeholders (see interview schedule, question nine). The questions ask:

Question nine: Does the organisation effectively communicate the value of outsourcing to all the stakeholders?

Data emanating from FTF interviews revealed that organisation does not effectively communicate the value of outsourcing to all the stakeholders. As one of the respondent noted (R6):

"There is a lack of effective communication between the Finance and the Supply Chain Management department."

Similarly, emerging from FGDs is concern with miscommunication with various units and planning. As one respondent noted (R8):

"The organisation is not effectively communicating the value of outsourcing to all the stakeholders due to poor planning and miscommunication between various stakeholders."

Emerging from the data are concerns around outsourcing challenges. There are challenges of poor procurement practices and unethical practices such as corruption, fraud and inflated pricing, lack of access to highly skilled service providers, poor planning and miscommunication. Thus, the researcher grouped these under the "Theme Cluster 2: Challenges and Impact."

4.4.3. Test of Research Objective Three:

Research objective three was structured to ascertain the existing monitoring measures and their impact on the outsourcing functions in the South African public sector. The questions that provided answers for this research objective are question two, seven, ten, eleven and twelve (See Appendix A). These

questions were structured to ascertain the existing monitoring measures and their impact on the outsourcing functions in the South African public sector. The questions ask:

Question two: Is there a clearly defined strategy in the processes and procedures related to outsourcing and how do you rate them? Please explain.

The data emanating from FGDs indicated that there are concerns with efficiency and performance levels with outsourcing in the public sector. As one of the discussants noted (R9):

"There is a negative impact on the outsourcing of activities and productivity performance and efficiency levels is very slow due to political interference, inflated pricing and unethical practices in the procurement processes."

Similarly, emerging from the face-to-face discussions with one of the management concerns organisational efficiency and performance levels of outsourcing in the public sector. As one of the respondents noted (R4):

"Services conducted by suppliers are not up to standard due to the lack of accountable systems, poor procurement management practices and control in a public procurement process usually leads to the wastage of government resources. Moreover, the department fails to choose activities that will be outsourced. Moreover, the department is struggling to comply with South African National Treasury (2003) standard procedure, which specifies that a tender process should take about 90 days to be processed. This type of inefficiency jeopardises our daily operations and strategic planning of our office."

The researcher probed with a question aimed at determining how the department ensures that outsourced programs are unique and have higher value than alternatives (see interview schedule, question seven).

Question seven: How does a department ensure that outsourced programs and services are unique and have higher value than?

Data from the FGDs indicate that outsourced programs and services are not unique and do not add value to the organisation. As one of the discussants noted (R8):

"Service providers offer poor services at an inflated price and existing monitoring measures are not good enough to hold suppliers accountable."

Similarly, the FTF interviews indicate that outsourced programs and services do not add value to the organisation. As one responded noted (R6):

"Outsourcing in any public sector does not add value due unethical practices in procurement processes."

The researcher asked a question to evaluate how the organisation measures results and how they identify areas for improvement (see interview schedule, question ten). The question asks:

Question ten: How does the organisation measure results? Do they actively identify areas for improvement?

Data from FTF interviews indicates that the department has control measures in place to ensure that potential service providers ensure delivery of quality service. As one responded noted (R6):

"Appointed suppliers sign a service level agreement with the specifications to ensure they deliver quality service at the right time."

The FGD discussions also reveal concerns with compliance and background checks, such as tracing of references on the potential suppliers. As one discussant noted (R6):

"The department is failing to evaluate past performance of potential suppliers to ensure that the selected suppliers can deliver quality services."

The researcher probed with a further question to determine how the department maintains client satisfaction (See interview guide, question eleven).

Question eleven: How does the department ensure that client satisfaction is maintained at a high level?

The FTF interviews indicated no concern with the level of client satisfaction. As one of the respondents noted (R7):

"Clients are happy with our services but poor control, poor monitoring and evaluation mechanisms of service providers need to be looked at so we can continue to maintain high levels of client satisfaction".

While probing during the FGD, one of the discussants revealed concerns with the quality of the service provided to customers. As one the discussant noted (R8):

"The quality of our service has decreased in past years due to unethical practices of some of our staff members and service provider and it has negative implications on the image of our department."

Lastly, the researcher probed further with a question to determine the rate and impact of control measures on employee's attitudes and behaviours (See interview guide, question twelve).

Question twelve: How do you rate the impact of the control measures on attitudes and behaviours of the employees?

The data emanating from FGD discussions indicated that there are concerns with poor control measures and poor enforcement of laws and regulations that deals with corruption in procurement processes. As one of the discussants noted (R6):

"There are so many corruption cases in the public sector but there is a zero conviction rate so the corrupted employees have a negative attitude about the laws and regulations of South Africa."

Similarly, emerging from the face-to-face discussions with one of the management, there are concerns about poor control on procurement processes and their management which leads to corrupt employees doing as they please. As one of the respondents noted (R5):

"There are poor control measures and all the corrupt officials are protected by senior management."

Emerging from the data are concerns around poor procurement management practices and controls, lack of accountability systems, poor compliance, controls and enforcement of laws or regulations that deal with unethical practices such as political interference, inflated pricing, fraud and corruption, thus the researcher grouped these under the "Theme Cluster 3: Monitoring Measures and their Impact."

4.5. Discussion of research findings

Research findings refer to a process of arranging and reporting the results of the study conducted, based on the methodologies applied in the gathering of information (Sekaran and Bougie, 2016). Discussions on findings are presented as per each research objective.

4.5.1. Findings based on barriers affecting outsourcing

The findings showed that the barriers facing outsourcing in the South African public sector include: inefficiency; lack of proper planning and incompetency in the SCM unit; human resource incapacity; shortage of staff and lack of proper qualifications; and insufficient budget allocation in a procurement unit. Moreover, the employees in the procurement unit are not paid well.

The findings from the tested research objective one indicate concerns emerging from two variables, namely, inefficiency and shortage of resources.

Findings from focus groups discussion show that the supply chain unit is suffering from a shortage of resources such as lack of proper SCM skills, knowledge and competency, shortage of qualified

personnel, shortage of budget, and lower salaries due to implementation of budget cuts policies by the South African government. Kingpin (2015) proclaimed that the public sector practices outsourcing as a cost-cutting measure without considering the impact of the outsourced service on the overall process flow.

Data from management shows that there is high inefficiency in procurement units, which is caused by lack of proper planning and budget cuts. A study conducted by Taponen (2017) confirmed that the purchasing unit did not achieve efficiency due to poor skills and incompetency of the officials working in the procurement unit. Moreover, the management agrees that they do not have proper SCM qualifications and skills but claim that they have years of experience in managing different units in the public sector. Furthermore, the management is not happy with their salaries. Gurría (2016), highlighted that lack of proper SCM skills, knowledge and competencies hinder consistent procurement practices and application of department policies into practices. The study conducted by Mamiro (2010) asserted that inadequacy of skills and qualifications in any unit increases inefficiency and interrupts organisational growth.

Given these finding, it is established that the department is inefficient and lacks proper planning, and there is incompetency in the SCM unit, human resource incapacity, shortage of staff and a lack of proper qualifications, and insufficient budget allocation in a procurement unit. It is essential, according to Hausman *et al.* (2008), to address the skills shortages with a combination of long-term educational reforms and short term solutions, such as easing up of immigration regulations for highly qualified individuals who can facilitate the rapid accumulation of skills.

4.5.2. Findings the effects of outsourcing on business operations

The findings showed that the effect of outsourcing on business operations are: poor procurement practices and unethical practices such as corruption, fraud and inflated pricing; lack of access to highly skilled service providers; poor planning; and miscommunication.

The findings from the tested research objective two, indicate concerns emerging from four variables, namely poor procurement practices, unethical practices, poor planning, and miscommunication

Findings from management interviews reveal that poor procurement management practices has affected the organisation's efficiency. SCM Review (2015) reveals that the public sector outsourcing is facing high levels of concern about current procurement practices. The findings agree with those of Mnguni (2012), which identify the challenges in the appointments of service providers, uniformity in procurement processes and good governance in the public sector. SCM legislative frameworks are used

to guide procurement practices in ensuring fairness, equity, transparency, competitiveness and costeffectiveness (National Treasury, 2014). Moreover, unethical practices in procurement processes are
amongst the issues that create a negative impact on the business operation, and they have reduced
productivity levels in the department (National Treasury, 2017). Some of the major challenges in the
public procurement processes, according to Badenhorst-Weiss (2012), are fraud, inflated pricing,
political interference and corruption. Research done by De Lange (2011) states that about 20% of the
budget in the public sector is wasted each year due to unethical practices associated with procurement
and the mismanagement of procurement processes.

Data from focus group discussions reveals the organisation's lack of efficiency due to poor planning and the reduced productivity and efficiency of the organisation. Smart procurement (2011) identified poor planning in a procurement unit as a major barrier that affects the organisation. Badenhorst-Weiss (2012) identified inadequate planning in the public sector as a key challenge that increases inefficiency in daily operations of the public organisation. Moreover, the procurement unit is failing to have proper communication with all relevant stakeholders. According to Mamiro (2010), poor communication between stakeholders has a negative impact on the organisation's procurement. In 2003, the South African Cabinet adopted a SCM policy to replace outdated procurement and provisioning practices. The aim was to implement a SCM function across all spheres of government, which would be an integral part of financial management and would conform to international best practices (National Treasury, 2005). O'Sullivan (2013) noted that the success of outsourcing solely depends on an enabling environment established at national level.

Given these finding, it is established that the department is struggling to access qualified service providers due to poor procurement practices and unethical practices in a procurement unit, and the procurement unit is struggling to communicate with other stakeholders.

4.5.3. Findings based on research objective three

The findings related to the monitoring measure and its impact show poor monitoring and evaluation controls, lack of accountability systems, poor compliance, and poor enforcement of SCM laws or regulations that deal with unethical practices in procurement processes.

The findings from the tested research objective three indicate concerns emerging from three variables, namely, poor monitoring and controls, lack of accountability systems, and poor compliance and enforcement of South African law when dealing with unethical practices.

Data from management reveals that the organisation has a poor monitoring and evaluation mechanism. The findings agree with those of Ambe and Badenhorst-Weiss (2012), which reveal that inadequate monitoring and evaluation in procurement processes are a key challenge in South African public sector. Bent (2014) notes that the public sector only focuses on the implementation phase of procurement and legislative compliance, and thus does not fully explore the performance phase of procurement processes. Selemi (2015) reveals that the South African government loses a lot of money due to a lack of capacity to monitor contractors. According to Acevedo et al. (2010), a strong monitoring and evaluation systems ensures the means to provide sound governance and accountable public policies. Moreover the study reveals that the department lacks accountability systems. Accountability is a central pillar to public sector outsourcing (Porteous and Naudé, 2012). According to Jeppesen (2010), without accountable and transparent systems, the vast state resources run the danger of being entangled with wastage and corruption. Furthermore, Ambe and Badenhorst-Weiss (2012) reveal that without proper accountability mechanisms, public entities will not account for their actions, which easily opens up the risk of increased fraud and corruption. Moreover, the study shows that the department has SCM compliance issues. According to Watermeyer (2011), SCM is guided by a number of related regulations and policies but compliance with these policies and regulations in a public sector is still a problem. Training and continuous capacity building are needed for effective monitoring and evaluation processes (Dlamini and Migiro, 2016).

Findings from the focus group discussions shows that the organisation has issues of enforcement of laws that deal with corruption and fraud. Munzhedzi (2016) explains that a lack of proper application of disciplinary procedures is the main driver of fraud and corruption. The application of corrective measures to these concerns increases costs in the organization.

The study adopted two theories, namely the Relational View Theory (RVT)/ Resource-Based View Theory and Core Competency Theory (CCT). According to Alvarez-Suescun (2010), RVT helps an organisation to achieve competitive advantage and efficiency through organisation capabilities, resources and competencies. The study's findings show that in the organisation, there is a lack of proper planning, incompetency in the SCM unit, human resource incapacity, shortage of staff, unethical practices, lack of proper qualifications and insufficient budget allocation in a procurement unit. Adoption of RVT can help the organization to achieve efficiency through evaluation, streamlining and accountability in the use of its internal resources as well as external resources and relational networks.

CCT is used as a tool to determine the success of an agreement with the suppliers. The findings of the study show poor procurement practices, poor monitoring and evaluation controls, lack of accountability

systems, poor compliance and poor enforcement of SCM laws or regulations that deal with unethical practices in procurement processes. CCT can be used as a knowledge-sharing and knowledge application instrument to ensure the success of an outsourcing process (Nasiopoulos, Sakas and Vlachos, 2014). Moreover, this means that the department can apply CCT when they determine the success of an agreement with suppliers.

Given these finding, it is established that the department is struggling with monitoring and control of service providers. Moreover, the department does not have proper accountability systems to ensure that they hold suppliers accountable for their actions. In addition, the department has compliance issues. Furthermore, the department is failing to apply or enforce South African law when dealing with unethical practices in the procurement processes, and it has cost tax payers a lot of money.

4.6. Conclusion

This chapter provided the presentation and analysis of data, response rate and a test of the research objectives. Furthermore, it analysed the research findings. The next chapter provides a summary, conclusions and recommendations of the study.

Chapter Five

Summary, Recommendations and Conclusion

5.1. Introduction

This chapter presents the summary of all the chapters, and recommendations and conclusions based on the data analysed in the previous chapters. Chapter one provided an overview of the research under study. Chapter two served as a review of literature on the concept of outsourcing in the public sector. Chapter three covered the research design and methodology of the study. Chapter four discussed the presentation and analysis of the data collected. Chapter five provides the summary of chapters, recommendations and the concluding remarks of the research.

5.2. Summary of chapters

This section provides a synopsis of all the chapters presented in this dissertation.

5.2.1. Summary of chapter one

Chapter one provides an introduction, the background to the study and the case study organisation. In addition, the chapter provides information on the statement of the research problem, the research objectives and questions in line with the research problem and the objectives of the research problem. These questions aim to "identify the barriers facing outsourcing of functions in the South African public sector." Three questions were derived for the study to address the identified research problem. The research questions are:

- What are the barriers facing outsourcing of activities in the South African public sector?
- To what extent do challenges of outsourcing affect the organisation's operations?
- What are the monitoring measures in place and how do they affect the outsourcing activities in the South African public sector?

It was anticipated that a feasible solution would be found to control the barriers affecting the outsourcing of functions in the South African public sector. In addition, the chapter introduced the literature review, provided a summary of the research methodology, and presented the significance, justification, contribution and limitations of the study, and a summary of each chapter.

5.2.2. Summary of chapter two

Chapter two served as a review of the current literature on the concept of outsourcing in the public sector. This chapter explored existing literature and provided in-depth knowledge on the area of outsourcing. The information provided covers a historical perspective on outsourcing and theoretical models underpinning the concept of outsourcing. In addition, this chapter addressed emerging trends

and factors driving outsourcing from a public sector perspective. The chapter addressed outsourcing activities, a perspective from SASSA and outsourcing challenges in the public sector, as well as the processes and procedures for outsourcing. Lastly, the dimensions and classification of outsourcing were discussed.

5.2.3. Summary of chapter three

Chapter three addressed and outlined the research methodology used in data collection for the study. The chapter covered the research design and paradigms, approaches to the research study and the research instruments adopted for the study. The chapter concludes by outlining the research population, the sampling technique used in the study, and reliability and validity of the study.

5.2.4. Summary of chapter four

Chapter four covered data presentation and analysis. It provided the presentation and analysis of data, response rates and the test of research objectives. In addition, the discussion of research findings was analysed.

5.2.5. Summary of chapter five

This chapter is the last chapter. It presents the summary of all the chapters, summary of research findings and the study recommendations. In addition, this chapter and describes ethical considerations.

5.3. Summary of findings on research objectives

The aim of this study was to examine the barriers in the effective use of outsourcing in the public sector. This section discusses all key findings from research objectives one, two and three.

5.3.1. Summary of key findings on research objective one

The key finding of research objective one is that there is inefficiency, lack of proper planning and incompetency in the SCM unit, human resource incapacity, shortage of staff and lack of proper qualifications, and insufficient budget allocation in a procurement unit. Moreover, the employees in the procurement unit are not paid well. In addressing the shortage of skills and lack of qualifications, the study recommends that the department needs to offer long term educational reforms, such as offering bursaries to young permanent and non-permanent employees, and that it hires highly skilled immigrants to facilitate the rapid accumulation of skills.

5.3.2. Summary of key findings on research objective two

The key finding of research objective two is that the department is struggling with poor procurement practices and unethical practices such as corruption, fraud and inflated pricing, lack of access to highly skilled service providers, poor planning and miscommunication.

5.3.3. Summary of key findings on research objective three

The key finding of research objective three is that the department is struggling with poor monitoring and evaluation controls, lack of accountability systems, poor compliance and poor enforcement of SCM laws or regulations that deal with unethical practices in procurement processes.

5.4. Study recommendations

This study highlighted challenges for which the researcher was able to offer recommendations for future studies. Some of these challenges were highlighted in chapter one of this dissertation. Based on the challenges encountered, the researcher recommends the following:

5.4.1. General recommendations from the study

Based on the research carried out, the researcher has found that the department is suffering from inefficiency, shortage of resources and unethical practices. This section discusses all recommendations from research objectives one, two and three.

5.4.1.1 Recommendation on research objective one

The department is suffering from inefficiency, lack of proper planning, incompetency and shortage of resources. The SCM procurement unit should have detailed procurement plans and the finance unit should apply for more funding from the National Treasury. The HR department should have knowledge and understanding of the skills and competencies required in SCM. Since the procurement unit does not have adequate staff, the HR department needs to hire more SCM staff with high ethical standards; and they could hire them on a contractual basis since the department is still struggling with budget cuts adopted by National Treasury. This will ensure that the procurement department is fully staffed at all time.

5.4.1.2 Recommendation on research objective two

The department is struggling with poor procurement practices and unethical practices, lack of access to highly skilled service providers, poor planning and miscommunication. The study recommends that the department work closely with law enforcement agencies and the criminal justice system to eliminate unethical practices in the public sector. Moreover, addressing the poor procurement practices in the public sector, the study recommends that the department needs to look for skilled individuals from

around the globe. The study recommends that the department needs to go for team building to ensure that they communicate.

5.4.1.3 Recommendation on research objective three

The department is struggling with poor monitoring and evaluation controls, lack of accountability systems, poor compliance and poor enforcement of SCM laws or regulations that deal with poor monitoring and evaluation controls and unethical practices in procurement processes. The department is struggling with monitoring and evaluation of service providers. Therefore, the study recommends the adoption of consequence management in the public sector and the appointment of someone who will monitor the performance of existing suppliers.

5.4.2. Recommendations for future academic research

Based on the research carried out, the researcher has found areas that need to be improved in case of future studies. Future researchers should try to use a larger sample size instead of the small sample size used into this study, in order to explore the opinions of more individuals with the aim to have a much more in-depth conclusive result. Furthermore, future researchers can undertake this kind of study by using more than one department or unit as the basis of collecting data in order to identify the barriers facing the outsourcing of functions in the South African public sector.

5.4.3. Recommendations for policy-makers

It is important that policy-makers and the government ensure that they increase the budget in the public sector to increase efficiency and effectiveness. To deal with the challenges and barriers of outsourcing, such as fraud and corruption, government needs to strengthen law enforcement agencies and the criminal justice system to ensure that there are thorough investigations that lead to the arrest of all perpetrators of fraud and corruption in procurement processes in the public sector. To access highly skilled officials, the government needs to ease up on immigration regulations for highly qualified foreign individuals who can facilitate the rapid accumulation of skills. The government should monitor procurement processes in the public sector to stop the non-compliance of public officials and ensure adherence to all the National Treasury policies and regulations. Furthermore, there should be training and skills development for all the employees in SCM. The study recommends that the government needs to adopt consequence management for all the officials who fail to comply with procurement laws and regulations. This dissertation can be used as a starting point for data collection. However, there should be more studies to increase the pool of information that can be used to draft procurement policies.

5.5. Ethical considerations

According to Creswell (2014:92), "ethics is defined as a set of rules or values, norms and principles of an individual that govern a particular person's moral behaviour". All the necessary steps for obtaining ethical permission from the University of KwaZulu-Natal research office were taken. Ethical approval for this research was obtained from the University of KwaZulu-Natal's Ethics Committee. Prior to data collection, all respondents and discussants were informed about the purpose of the study, that their participation was voluntary, and they were requested to complete an informed consent form (refer to Appendix B). Furthermore, throughout the course of the study, all the rules and regulations were adhered to in ensuring that this study maintains highest levels of privacy, confidentiality and anonymity (see Appendix B). Furthermore, in ensuring anonymity or protection of the identity of the respondents, the records will be maintained by the School of Management, IT and Governance, UKZN.

5.6. Conclusion

The aim of the study is to identify the barriers facing the outsourcing of functions in the South African public sector. Findings show that there are inefficiencies, lack of proper planning and incompetency in the SCM unit, human resource incapacity, and shortage of staff, lack of proper qualifications and insufficient budget allocation in a procurement unit. Moreover, the department is struggling with poor procurement practices and unethical practices, such as corruption, fraud and inflated pricing, lack of access to highly skilled service providers, poor planning and miscommunication. Further, the department is struggling with poor monitoring and evaluation controls, lack of accountability systems, poor compliance and poor enforcement of SCM laws or regulations that deal with unethical practices in procurement processes. The department is struggling with monitoring and evaluation of service providers.

The literature review and primary data collected in the public sector have been able to investigate three questions surrounding the barriers facing the outsourcing of functions in the South African public sector. According to Vintar and Stanimirovic (2011), the outsourcing of functions should increase efficiency and effectiveness and can help organisations to focus on the core business, Indeed, as the present study has discussed, that was a key intention behind outsourcing in the public sector in South Africa. However, the present research with management and staff of one SASSA office in KZN shows otherwise, and in some instances confirms Munzhedzi's (2016) argument that outsourcing in the public sector does not yield positive results due to corrupt practices in the outsourcing processes. Government will continue to lose money if it does not deal harshly with perpetrators. Government is making an effort to improve outsourcing processes and ensure that value for money is achieved in all public entities but the challenges of unethical behaviour, such as fraud and corruption in the public sector, is still on the rise (National Treasury, 2018). In closing, outsourcing itself is not a panacea that guarantees efficiency,

cost saving, quality or improved service in the public sector, but with rigorous implementation of the necessary controls and accountability measures, budgets to meet its objectives, and the appointment of highly skilled experts, it can be effective in resourcing the public sector to serve the people with integrity, and with quality service.

References

- Acedo, F.J., Barroso, C., and Galan, J.L. 2006. The resource-based theory: dissemination and main trends. *Strategic Management Journal*. 27: 621-636.
- Acevedo, G.L., Rivera, K., Lima, L. and Hwang, H. 2010. Challenges in monitoring and evaluation: an opportunity to institutionalize M and E systems. Washington, DC: The International Bank for Reconstruction and Development.
- Aguilera, R.V., Rupp D.E, Williams C.A and Ganapathi, J. 2007. Putting the S Back in Corporate Social Responsibility: A Multilevel Theory of Social Change in Organisations. *The Academy of Management Review*. (Online). 32(3): 836-863. Available from: https://www.jstor.org/stable/20159338?seq=1#metadata_info_tab_content. (Accessed 14 January 2018).
- Ambe, I.M. 2009. An exploration of supply chain management practices in the central district municipality. *Educational Research and Review* 4(9): 427-435.
- Ambe, I.M. and Badenhorst-Weiss, J.A. 2011. An exploration of public sector supply chains with specific reference to the South African situation. *Journal of Public Administration* (online), 46(3): 1100–15. Available from:https://journals.co.za/content/jpad/46/3/EJC51828. (Accessed 15 June 2018).
- Ambe, I.M. and Badenhorst-Weiss, J.A. 2012. Procurement challenges in the South African public sector. *Journal of Transport and Supply Chain Management* (online), 6(1): 242-261. Available from: https://doi.org/10.4102/jtscm.v6i1.63. (Accessed 20 April 2018).
- Ambuj, K, Anurag S, and Peeyush T. 2012. Supply Chain Elements that affects the retail sales of electronic goods in the Oman region. *Arabian Journal of Business Management Review* (online), 1(8). Available from:https://www.researchgate.net/publication/266417665_Supply_Chain_Elements_that_affectsthe_retail sales of electronic goods in the Oman region. (Accessed 25 December 2018).
- Andersson, D., Fransson, D and Rehme, J. 2006. Driving forces for outsourcing: A study of wood product manufacturing firms. Ipsera Conference on driving forcing of outsourcing. Sweden, 12-15 January 2006. Linkoping: Linkoping University, 24-27.
- Aterido, R., A. Hlatswayo, D. Pieterse, and A. Steenkamp (2019). 'Firm Dynamics, Job Outcomes and Productivity: African Formal Businesses, 2010-14'. World Bank Policy Research Working Paper 8788, Available at: http://documents.worldbank.org/curated/en/140241553174981836/pdf/WPS8788.pdf (Accessed: 16 May 2020).
- Arrowsmith, S. 2010. Horizontal policies in public procurement: a taxonomy. *Journal of Public Procurement* (online).10: 149-186. Available from: https://doi.org/10.1108/JOPP-10-02-2010-B00. (Accessed 12 April 2018).
- Arya, B and Lin, Z. 2007. Understanding collaboration outcomes from an extended resource-based view perspective: The roles of organisational characteristics, partner attributes, and network structures. *Journal of Management*, 33(5): 697-732.
- Ashenbaum, B., Maltz, A and Rabinovich, E. 2005. Studies of trends in third-party logistics usage: what can we conclude? *Transportation Journal*. 45(3): 281-286.
- Babbie, E. 2004. The practice of social research. 2nd edition. Cape Town: Thompson Learning.

- Badenhorst-Weiss, J.A., 2012. Procurement challenges in the South African public sector. Master's degree, University of South Africa.
- Barnett, M., 2007. Tarredand: Exploring Interdependence in the Volatility of Stock Returns, *Corporate Reputation Review* 10(1): 3–21.
- Barney, J. B., Ketchen, D. J., Jr and Wright, M. 2011. The future of resource-based theory: Revitalization or decline? *Journal of Management*, 37, 1299–1315.
- Barney, J.B, 1991. Firm resources and sustained competitive advantage. *Journal of management* (online), 17(1): 99-120. Available from: https://www.business.illinois.edu/josephm/BA545_Fall%202015/Barney%20 (1991).pdf. (Accessed 1 June 2018).
- Barney J.B., Hesterly W. 1996. "Organisational Economics: Understanding the Relationship between Organizations and Economic Analysis", in Clegg S.R., Hardy C., Nord W.R. (Eds.) Handbook of Organization Studies, Sage Publications, London.
- Barthélemy, J. and Quélin, B.V. 2006. Complexity of Outsourcing Contracts and Ex Post Transaction Costs: An Empirical Investigation. *Journal of management studies* (online), 43:1775-1779. Available: https://onlinelibrary.wiley.com/doi/abs/10.1111/j.1467-6486.2006.00658.x (Accessed 18 January 2020).
- Barreto, I. 2010. Dynamic capabilities: A review of past research and an agenda for the future. *Journal of Management*, 36, 256–280.
- Bent, R.A. 2014. The significance of supply chain management with regard to the attainment of value and strategic objectives for municipalities within South Africa: A case study. Stellenbosch University. Stellenbosch, South Africa.
- Belcourt. M. 2006. Outsourcing benefits and risks. *Human resource management review* (online), 16: 269-279. Available from: https://www.sciencedirect.com/science/article/abs/pii/S1053482206000234.(Accessed 03 January 2019).
- Bera, S.P. 2017. The concept of outsourcing. *Journal of Computer Engineering* (Online), 2278-8727. Available from: https://www.bera.ac.uk/event/bera-conference.(Accessed 15 November 2018).
- Bhagattjee, P. and Hofmeyer, C.D. 2009. South African outsourcing laws. (Online). Available from: hhppt://www.practicallaw.com/outsourcinghandbook.html (Accessed 03 March 2019).
- Bolton, P. 2009. Government procurement and policy promotion in South Africa, in Procurement from developing countries and economies in transition. *Journal of public procurement*, 6(3): 193-217.
- Bhorat, H. and Cassim, A. 2014. South Africa's Welfare Success Story II: *Poverty Reducing Social Grants* (Online) Available from http://www.brookings.edu/blogs/africa In-focus/posts/2014/01 /27 -south-africa-welfare-poverty-bhorat (Accessed 14 June 2018).
- Boateng, D. 2008. Government service delivery lies in supply chain management, not centralized procurement, smart procurement. Available from: www.smartprocurement. co.za (Accessed 10 February 2018).

- Bolton, P.B. 2006. Government procurement as a policy tool in South Africa. *Journal of Public Procurement* (online), 6(3): 193–217. Availabe from: http://www.ippa.org/jopp/download/vol6/issue-3/Article1_Boltol.pdf. (Accessed 13 February 2018).
- Barney, J. and Hesterly, W. 2010. *Strategic management and competitive advantage: concepts and cases.* 3rd ed. New Jersey: Prentice Hall.
- Bovis, C.H. 2012. "Risk and Public-Private Partnerships." *European Procurement & Public Private Partnership Law Review.* 7(1): 44-56.
- Brewster, C., Carey, L., Grobler, P., Holland, P., and Warnich, T. 2000. Contemporary issues in human resource management: gaining a competitive advantage. Oxford University Press Southern Africa.
- Bryman, A. 2008. Social research methods. 2nd Ed. Oxford: Oxford University Press.
- Bryman, A. and Bell, E. 2011. Business research methods. 3rd ed. Oxford: Oxford University Press.
- Business Analytics. 2015. What are the advantages of purposive sampling? *The international journal of higher education and planning* (Online), 15: 166. Available from: http://www.enkivillage.com/purposive-sampling.html (Accessed 15 October 2018).
- Cacciatori, H. and Jacobides, M.G. 2005. The dynamic limits of specialization: vertical integration reconsidered. *Strategic and International Management Area*. Volume/20-26.
- Cha, H.S., Pingry, D.E and Thatcher, M.E. 2008. 'Managing the Knowledge Supply Chain: An Organisational Learning Model of Information Technology Offshore Outsourcing.' *Management Information Systems Research Center, University of Minnesota*, 32(2), 281-306.
- Charm, N. and Gottarna J.1990.The *human resource factors in logistics*. 1st ed. Gower: Gower publishing limited.
- Chuck, C. 2015. Different types of outsourcing: strategic way of outsourcing. *International journal of economic practices and theories* (online). 15:123-160. Available from: https://www.forbes.com/sites/chuck (Accessed 2 September 2018).
- Close, D. 2004. What is a 3PL? How the firm decides if we need one? Logistics quarterly .10(4). Available from: https://www.iq.ac/issues/nov2004/articles/articles01.html (Accessed 10 October 2017).
- Cook, T. 2014. Australian government moves to outsource key social services (online). Canberra: Department of Human services. Available from: http://www.wsws.org/en/articles/2014/08/27/media27.html (Accessed 07 May 2018).
- Corbett. M. 2004. *Outsourcing revolution: why it makes sense and how to do it right.* 2nd ed. Recuperado de Dearborn trade publishing, 176-190.
- COSATU. 2011. Draft position paper for the public service summit (3rd version) (Online) Available from http://www.cosatu.org.za/docs/policy/2002/02priva.htm (Accessed 07 May 2018).
- Creswell, J.W. 2014. Research design: qualitative, quantitative, and mixed methods approaches. 5nd ed. London: Sage.
- Cuzak, G. 2008. Outsourcing of marketing activity. Master's degree, Catholic University of Eichstaff
- Davies, R. 2003. Corporate Social Responsibility: What can we learn from the Stakeholders?

- Social and behavioural journal (online), 13:327-337. Available from: https://www.sciencedirect.com/science/article/pii/S1877042812051154.(Accessed 08 June 2018).
- Davis, K.1973. The Case for and Against Business Assumption of Social Responsibilities. *Academy of Management Journal* 16(2), 312–322.
- De Lange S. 2011. Irregular State Expenditure Jumps 62%. *Business Day*, 03 April: 07. (Accessed on 20 October 2018).
- De Vaus, D.A. 2001. Research design in social research. London. SAGE.
- Denning, S. 2013. What went wrong at Boeing? *Strategy and Leadership journal* (Online), 41:3 Available from: http://www.forbes.com/sites/steveden (accessed 07 May 2018).
- Denzin, N. 2010. On elephants and gold standards. *Qualitative Research*, 10(5):269–272.
- Department of Social Development. 2018. Media statement, (online). London. Department of Social Development. Available from: http://www.dsd.gov.za. (Accessed 01 October 2018).
- Department of Treasury.2016. Fruitless and wastefully expenditure mid-term presentation, (online). London. Department of Treasury. Available from:http://www.National Treasury.gov.za/department/department.html (Accessed 03 March 2017).
- Devi, P.S. 2017. Research Methodology: A Handbook for Beginners. 2nd ed. Chetpet: Notion Press
- Dinu, A.M., 2015, 'The risks and benefits of outsourcing', Knowledge Horizons. Economics 7(2), 103–104.
- Diez-Vial, I., Alvarez-Suescun, E. 2010. Geographical Agglomeration as an Alternative to Vertical Integration. *Review of Industrial Organisation* (online), 36:373–389. Available: https://doi.org/10.1007/s11151-010-9251.pfd (Accessed 4 May 2019).
- Doug P., 2018, *global outsourcing* (online), 32:161-170. Available from: https://www.deloitte.com/content/dam/Deloitte/global/Documents/Process-and-Operations/gx-2018-global-outsourcing-survey-executive-summary.pdf (Accessed 2 September 2018).
- Durrheim, K. and Pienter, D. 2006. *Collecting quantitative data: sampling and measuring*. 2nd ed. Cape Town: UCT Press.
- Dhall, S.C., 2013, Outsourcing, Delhi Press, New Delhi.
- Dlamini, P.C. and Migiro, S.O. 2016. Performance Monitoring and Evaluation Systems in the South African Local Government. *Journal of Public Administration*, 51 (3): 376-390.
- Edmiston, D. 2011. The shifting balance of private and public welfare activity in the UK, 1979 to 2007. *Centre for Analysis of Social Exclusion*, 15:125-129. Available from: https://www.core.ac.uk/download/pdf/8791899.pdf (Accessed 15 May 2018).
- Eyaa, S. and Oluka, P.N. 2011. Explaining non-compliance in public procurement in Uganda. *International Journal of Business and Social Science*, 2(11): 35 44.
- Fawcett, S.E., Ellram, L.M. and Ogden, J.A. 2012. *Supply chain management from vision to implementation*. 2st ed. Upper Saddle River: Pearson Prentice-Hall.

- Feeny D., Lacity M. and Willcocks L. 2005. Taking the measure of outsourcing providers. *Management Review*, 46 (3): 41-48.
- Fink, A. 2002. *The Survey Kit*. 2nd ed. London. SAGE Publications.
- Gerrard, D.V, Gideon N. and Wesley. 2012. *Strategic logistics management: a supply chain management approach*. 3rd ed. Pretoria: Van Schaik publishers.
- Gurría, A. 2016. Public procurement for sustainable and inclusive growth: *Enabling reform through evidence and peer reviews* (online). Available:http://www.oecd.org/gov/ethics/PublicProcurementRev9.pdf (accessed on 21 May 2020).
- Grabman, G. M. 2005. *The Non-profit Handbook: Everything You Need to Know to Start and Run Your Non-profit Organisation*. 5th ed. Pennsylvania: White Hat Communications.
- Gunasekaran, A. 2007. Knowledge management in 21st century manufacturing: Century manufacturing, *International Journal of Production Research*, 45:11, 391-418.
- Gupta, S., Armstrong, K. and Clayton, Z. 2011. Social media. *Harvard Business School marketing journal* (online), 2(2):510-095. Available from: https://www.hbs.edu/faculty/Pages/item.aspx?num=38521.(accessed 11 November 2018).
- Gurría, A. 2016. Public procurement for sustainable and inclusive growth: Enabling reform through evidence and peer reviews, OECD: Available from: http://www.oecd.org/gov/ethics/PublicProcurementRev9.pdf (accessed on 21 April 2019).
- Hair, J.F., Celsi, M. W., Oritinau, D, J and Bush R. P.2013. *Essentials of Marketing Research*. 3rd ed. London: McGraw-Hill.
- Harland, C., Knight, L., Lamming, R. and Walker, H. 2005. "Outsourcing: assessing the risks and benefits for organizations, sectors and nations", *International Journal of Operations & Production Management*, 25 (9): 831-850.
- Hanks, J., Davies, H. and Perera, O. 2008. Sustainable public procurement in South Africa, *International Institute for Sustainable Development* (online), 3:23-32. Available from: https://www.iisd.org/pdf/2008/spp_south_africa.pdf (accessed 18 April 2018).
- Hausmann, R., B. Klinger, and R. Lawrence. 2008. 'Examining Beneficiation'. CID Working Paper 162. Boston: Harvard.
- Herath, T. and Kishore, R. 2009. Offshore outsourcing: risks, challenges and potential solutions, *Information Systems Management*, 26 (4): 312-326.
- Hess, D., Rogovsky, N and Dunfee, T.2002.The Next Wave of Corporate Community Involvement: Corporate Social Initiatives, *California Management Review* 44(2): 110.
- Iqhal, Z. and Dad, M. D. 2013. Outsourcing: A Review of Trends, Winners & Losers and Future Directions. *International Journal of Business and Social Science*, 4(8) (Suppl.): 91-106.
- Jensen, P.H. and Stonewash, R. E. 2004. The efficiency of public sector outsourcing contracts. *A literature review journal*, 29: 04-08.

- Jeppesen, R. 2010. Accountability in public procurement transparency and the role of civil society. *United Nations Procurement Capacity Development Centre*, (Online), 23: 235-304. Available from: http://www.unpcdc.org (Accessed on 7 March 2018).
- Joppe, M. 2000. The reliability and validity of qualitative and operational research paradigm. *Punjab Education Assessment Systems*, (Online), 4:1-59. Available from: http://www.researchgate.net/~mjoppe/rp.htm (accessed 02 February 2018).
- Kakabadse, A and Kakabadse, N. 2000. Critical Review outsourcing a paradigm shift. *The journal of management review*, 19(8) 670-701.
- Khalid, S. and Al Badi, S. 2017. The Dimensions of Marketing Mix Management. *Organisational Studies*.2 (1):136-142.
- Khatik, S.K. and Khanooja, G. 2013. Cost effectiveness of vehicles by outsourcing: A case study of City Circle Bhopal, M.P.M.K.V.V. Co. Ltd., *Indian Journal of Commerce and Management Studies* 4(2):37.
- Kingpin, K. 2015. When outsourcing is not a good idea. *Business resource journal*, (online), 2:21. Available from: https://www.streetdirectory.com/travel_guide/16669/outsourcing/when_outsourcing_is_not_a_good_i dea.html.(Accessed 07 July 2018).
- Kaplan, R. and Norton, D. 2005. "The Office of Strategy Management", *Harvard Business Review*, 83(10): 72-74
- Kaplan, R. and Norton, D. 2004. *Strategy Maps: Converting Intangible Assets into Tangible Outcomes*, Harvard Business School Press, Boston.
- Kotler, P., Armstrong, G. and Cunningham, P.H. 2005. *Principles of Marketing*. Toronto: Pearson Education Canada. Kotler, P. and Zaltman, G. 1971. Social Marketing: An Approach to Planned Social Change. *Journal of Marketing*, 35:3–12.
- Kotler, P and Lee, N. 2007. *Marketing in the Public Sector: A Roadmap for Improved Performance*. Warton: Pearson Education.
- Kruger, D., Piet, D. and Ramdass, R. 2007. *Operational Management*.2nd ed. Cape Town: Oxford University Press Southern Africa.
- Kraaijenbrink, J., Spender, J.-C., & Groen, A. J. 2010. The resource based view: A review and assessment of its critiques. Journal of Management, 36, 349–372.
- Kutsikos, K. and Mentzas, G. 2011. "A service portfolio model for value creation in networked enterprise systems", *Proceedings of service wave 2010 conference workshops*. Belgium, 6-9 July 2011. Ghent: Ghent University, 101-109.
- Kumar, R. 2008. Research Methodology: A step by step guide for beginners.5th ed. London. SAGE Publications.
- Lavie, D. 2006. The Competitive Advantage of Interconnected Firms: An Extension of the Resource-Based View. *Academic management review* (online). 31(3):124. Available: https://doi.org/10.5465/amr.2006.21318922. (Accessed 15 March 2019).
- Lawler, E., Fitz J and Madden, J. 2003. *Human resource business process outsourcing: Transforming how HR gets its works done*. San Francisco.CA: Jossey-Bass.

- Lien, T. B and Cian, F.S. 2006. Building marketing strategy for stated owned enterprises against private ones based on perceives of customer satisfaction and service quality. *Journal of Retail and Consumer Service*, 13 (1): 1-13.
- Lonsdale, C and Cox, A. 2000. The historical development of outsourcing: the latest fad. *Industrial Management and Data Systems*, 100(9): 444-450.
- Mahea, T. 2014. The Role of Marketing in Organisations (Link). Available from: https://www.linkedin.com/pulse/20140915080650-84228363-the-role-of-marketing-in-organisationsby-timothy-mahea (Accessed 15 December 2018).
- Mahlaba, P.J. 2004. Fraud and corruption in the Public Sector: An audit perspective. *Journal of Transport and Supply Chain Management*, 3(2):84–87.
- Mahmood, S.A.I. 2010. Public procurement and corruption in Bangladesh confronting the challenges and opportunities. *Journal of public administration and policy research*, 2(6):103.
- Magagula, M and Zondo, R. D. H. 2018. Uncovering the strengths and weaknesses of outsourcing core business deliverables: the case of selected state-owned enterprises in South Africa. The Journal for Transdisciplinary Research in Southern Africa (Online), 14(1):475. Available from: https://www.doi.org/10.4102/td.v14i1.475 (Accessed 22 May 2019).
- Martin, A. E. 2005. Social science research: Conception, methodology and analysis. Masters, Makerere University. (Accessed 04 August 2018).
- McCrudden, C. 2004. Using public procurement to achieve social outcomes. *Natural Resources Forum*, 257-267.
- McIvor, R. 2005. A practical framework for understanding the outsourcing process. *Supply Chain Management: An International Journal*, 5(1):22-36.
- McWilliams, A., Siegel, D.S. and Wright, P. M. 2006. Corporate social responsibility: Strategic implications. *Journal of Management Studies*, 43(1):1–18.
- Mehta, A., Armenakis, A., Mehta, N and Irani, F. (2006). Challenges and Opportunities of Business Process Outsourcing in India'. *Journal of Labour Research*, 27(3):24-338.
- Mercandetti T., 2018. Strategic outsourcing: benefits and risks of outsourcing (online). International Journal of Economic Practices and Theories. Available from: http://www.gruppolameridiana.com/en/strategic-outsourcing-benefits-risks (Accessed 2 September 2018).
- Mkhize, Z. 2004. Supply Chain Management Conference: Transforming Government Procurement System. *Paper presented at the Supply Chain Management Conference*, 22–23 Nov, Durban, Republic of South Africa.
- Michela, P. and Carlotta, M.C., 2011, 'Outsourcing strategies: How to formalize and negotiate the outsourcing contract', Annals of the University of Oradea: Economic Science 20(1), 276–287.
- Mokone, T. 2011. Irregular state expenditure. *Business Day*.11 December: 10. Available from: http://www.businessday.co.za (Accessed 10 February 2019).

- Mol, M. J. and Kotabe, M. 2009. Overcoming inertia: drivers of the outsourcing process. *The Fox School of Business journal*, (online). Available from: http://www.lrp.ac/ html (Accessed 11 October 2018).
- Monette, D. R., Sullivan, T. J. and Dejong C. R. 2011. Applied social research: a tool for the human services. Belmont: Brooks/ Cole.
- Monette, D.R., Sullivan, T.J. and DeJong, C.R. (2005). *Applied social research: A tool for the human services*. 6th ed. Toronto: Thomson Learning Inc.
- Mulgan, R. 2015. Transperancy and the performance of outsourcing government services. *The office of information commissioner journal*, (online), 5(3): 8-19. Available from: https://www.oic.qld.gov.au/__data/assets/pdf_file/0003/27444/paper-transparency-and-outsourced-government-services.pdf.ed (Accessed 03 April 2019).
- Mutiangpili J (2010). Government sector outsourcing transforming public service with outsourced IT services, (online). New York, USA: Tholons global offices. Http://Available: www.tholons.com. (Accessed 07 May 2018).
- Munzhedzi, P.H. 2016. "Fostering public accountability in South Africa: A reflection on challenges and successes". *The Journal for Transdiciplinary Research in Southern Africa*, 12(1): 7-8
- Munzhedzi, P.H. 2016. "South African public sector procurement and corruption: Inseparable twins?". Journal of Transport and Supply Chain Management, (10)1: 8-10
- National Treasury. 2019. *State of the Nation Address*. Online. Available from: http://www.national treasury/governance/ar.asp. (Accessed 08 March 2018).
- National Treasury. 2003. *Policy strategy to guide uniformity in procurement reforms processes in government*. Republic of South Africa.
- National Treasury. 2005. Supply chain management: A guide for accounting officers and municipal entities. Republic of South Africa.
- National Treasury, 2015. "Public sector supply chain management review". Pretoria. National Treasury. South Africa.
- Nasiopoulos, K., Dimitrios, Damianos P. Sakas, D.S. Vlachos . 2014. Modeling the Scientific Dimension of Academic Conferences. Procedia Social and Behavioral Sciences (pp. 467-477)
- Neeraj, K.S. 2016. Industry initiatives for green marketing in India. *Business economy journal*, 7: 192-198.
- Net 1. 2014. Home (Online). Available from: http://www.net1.com (Accessed 13 October 2018).
- Neuman, W. L. 2012. *Basics of social research: qualitative and quantitative approaches*. 2nd Ed. Boston: Pearson.
- Ngugi, K. and Nderitu, K.M. 2014. Effects of Green Procurement Practices on an Organisation Performance in Manufacturing Industry: Case Study of East African Breweries Limited. *European Journal of Business Management*, 2(1):341-352.
- Nordigarden, D., Rehme, J., Brege, D., Chicksand, D., and Walker, H. 2014. Outsourcing decisions a case study of a parallel production. *International Journal of operations and production management*, 34(8): 974-1002.

- Graf, M. and Mudambi, S.M. 2005. The Outsourcing of It-Enabled Business Processes: A Conceptual Model of the Location Decision. *Journal of International Management*, 11(2): 253-268.
- Odhiambo, W. and Kamau, P. 2003. Public procurement: Lessons from Kenya, Tanzania and Uganda. OECD Development Centre Working Paper No. 208.
- Osei-Tutu, E and Badu, E. 2010. Exploring corruption practices in public procurement of infrastructural project in Ghana. *International Journal of managing projects in business*, Fall:12:14
- O' Sullivan, S. 2013.Outsourcing procurement in the public sector. *Operations Services and Efficiency journal* (online), 8(3):33-40. Available from: https://www.adb.org/sites/default/files/publication/30337/outsourcing-procurement-public-sector.pdf, (Accessed 09 May 2018).
- O'Leary, Z. 2004. The essential guide to doing research. 3rd ed. London: Sage.
- Odhiambo, W. and Kamau, P. 2003. Public procurement: Lessons from Kenya, Tanzania and Uganda. *OECD Development Centre*, 3(3):208.
- Omar, K.2014. Tactical and strategic sourcing. *International journal of economic practices and theories* (online). Available from:https://blog.procurify.com/2014/08/26/tactical-sourcing-vs-strategic-sourcing-works-best-company/ Accessed 7 September 2018).
- Organisation for Economic Co-operation and Development, 2012. SIGMA support for improvement in governance and management. Available at: http://sigmaweb.org (accessed on 20 January 2018).
- Osborne, D and Gabler, T. 1992. *Reinventing Government: How the entrepreneurial spirit is transforming the Public Sector*. 1st ed. Addison: Wesley.
- Overseas Development Institute (2011). South Africa's cash social grants: Progress in Increasing coverage. Online. Available from: http://www.developmentprogress.org/sites/developmentprogress.org/files/southafrica.pdf. (Accessed 14 August 2018).
- Pauw, J.C. 2011. Public procurement and supply chain management. Study Guide, programme in Public Procurement and Supply Management. Pretoria, South Africa: Centre for Business Management, University of South Africa.
- Pauw, J.C. and Wolvaardt, J.S. 2009. Multi-criteria decision analysis in public procurement a plan from the South. Politeia, 28(1):66–88.
- Parikh, K. 2018. *Perfect competition* (online). Available from: http://www.letslearnfinance.com/advantages-and-disadvantages-of-perfect-competition.html (Accessed 15 October 2018).
- Parry, E. 2004. Offshore *outsource horse has left the barn* (online), Available:http://www.searchcio.techtarget.com/qna/0, 289202.sid19_gci913695, 00. hmtl. (Accessed 12 May 2018).
- Pensiero, N. 2017. In-house or outsourced public services? A social and economic analysis of the impact of spending policy on the private wage share in OECD countries. *International Journal of Comparative Sociology* (Online), 58 (4) 333–351. Available from: https://journals.sagepub.com/doi/10.1177/0020715217726837 (Accessed 22 May 2019).

- Pezullo, S and Rice, J. 2001. Designing a Successful Fund-Raising Campaign for Your NGO. International Youth Foundation. (Online). Available from:http://www.iyfet.org. (Accessed 23 April 2018)
- Potrafke, N. 2018. Does public sector outsourcing decrease public employment? Empirical evidence from OECD countries. Institute Leibniz Institute for Economic Research at the University of Munich (Online), No. 267. Available from: https://www.ifo.de/DocDL/wp-2018-267-potrafke-public-outsourcing.pdf (Accessed 22 May 2019).
- Porteous, E. and Naudé, F. 2012. Public sector procurement: The new rules (online). Available:http://www.apmp.org.za/public_ftp/ PPPFA%20presentation%20for%20web.pdf (accessed 03 April 2019).
- Prahaland, C.K. and Hamel, G. 1990. The core competence of the corporation. *Harvard Business Review*, 68(3): 79-91.
- Pribanic, E. 2018. Importance of Global Marketing Strategies for Businesses (online). Available from: https://www.techfunnel.com/martech/importance-of-global-marketing-strategies-for-businesses (Accessed 12 November 2018).
- *Purposive sampling works* (Online). 2015. Available from: http://www.enkivillage.com/purposive-sampling.html (accessed 27 October 2018).
- Quinn, J.B and Hilmer, E. 1994. Strategic outsourcing. Sloan Management Review, 35(4): 43–55.
- *Reducing Social Grants* (Online). 2018. Available from http://www.brookings.edu/blogs/africa (accessed 03 December, 2019)
- Regent institute of leadership and management. 2016. Impact of outsourcing practices in the public service (Online). London: Regen institute of leadership and management. Available from: https://www.pscbc.org.za/index.php? (Accessed 22 May 2019).
- National Treasury. 2018. *Policy strategy to guide uniformity in procurement reforms processes in government*, (online). London: National Treasury. Available from: http://www.treasury.gov.za/publications/strategic 20plan/strat20plan. (Accessed 25 March 2018).
- Richards, L. 2018. *Different Types of Market Systems* (online). Available from: https://smallbusiness.chron.com/5-different-types-market-systems-25818.html (Accessed 10 October 2018).
- Rumelt, R. P. 1997. Towards a strategic theory of the firm. Resources, firms, and strategies. *The resource-based perspective*, 131-145.
- Sani, A., Dezdar, A. and Ainin, A., 2013, 'Outsourcing patterns among Malaysian hotels', International Journal of Business and Social Science 4(9): 63–74.
- Sargeant, A. 2009. *Marketing Management for Non-profit Organisations*. 3rd ed. Oxford: Oxford University Press.
- SARB (South African Reserve Bank) (2018). 'Online Statistical Query'. Available at: https://www.resbank.co.za/Research/Statistics/Pages/OnlineDownloadFacility.aspx (accessed 8 March 2018).
- Saunders, M.N., Saunders, M., Lewis, P. and Thornhill, A. 2011. *Research methods for business students*. 5th Ed. New Delhi: Pearson.

- Schialfino, M. and Berkowitz, P. 2015. *Outsourcing in Mexico: Best practices* (Online). Available http://www.acc.com/accdocket/onlineexclusive/outsourcing-mexico.cfm (Accessed 07 May 2018).
- Sekaran, U. and Bougie, R. (2016). Research Methods for Business: *A Skill Building Approach*. 7th ed. United States of America: John Wiley and Sons, Inc.
- Selemi, M. 2015. Outcomes of outsourcing social grants payments services in the South African Social Services. Master's degree, North West University.
- Sen, S., Bhattacharya, C.B., and Korschun, D. 2006. The Role of Corporate Social Responsibility in Strengthening Multiple Stakeholder Relationship. *Journal of the Academy of Marketing Science*, 34(2):158–166.
- Sergi, V. and Hallin, A. 2011. Thick performances, not just thick descriptions: The procession nature of doing qualitative research. *Qualitative Research in Organisations and Management. An International Journal*, 6(2): 191-208.
- Serrat, O. 2010. Marketing in the public sector. Washington, DC: Asian Development Bank (online). Available from: https://digitalcommon.s.lr.cornell.edu/infl/144. (Accessed 23 December 2018).
- Sharma, A. and Loh, P. 2009. Emerging Trends in Sourcing of business services. *Business Process Management Journal*, (15)2: 163-165.
- Shaw, F.N. 2010. The Power to Procure: A Look inside the City of Austin Procurement Program. Master's Degree, Texas State University, United State of America.
- South Africa Social Security Agency (2015). Home (Online). Available from www.sassa.gov.za (Accessed on the 27 March 2018).
- South Africa. South African Social Security Agency Act No 09 of 2004. (2004). Pretoria. Government Printers.
- South Africa. Social Assistance Act No. 13 of 2004. (2004). Pretoria: Government Printers.
- South Africa. National Treasury. 2005. "Supply chain management: A guide for accounting officers and municipal entities". Pretoria, National Treasury.
- South Africa. National Treasury. 2003. "Policy strategy to guide uniformity in procurement reforms processes in government", Pretoria, National Treasury.
- Silverman, D. 2006. What is qualitative research? (Online). Available from: http://www.sagepub.com/upm-data/11254 Silverman 02.pdf (Accessed on 24 April 2018).
- Silverman, D. 2010. *Doing qualitative research*.3rd ed. Thousand Cape Town: Sage.
- Sinclair, M. 2007. A guide to understanding theoretical and conceptual frameworks. *Evidence based midwifery* (online), 5(2):39. Available from: http://www.doctoralmidwiferysociety.org/Portals/c8d3e3f8-9c01-4bf5-abd9-3fd6b4c510ae/marleneeditorialtheoreticlaframework.pdf (Accessed on 15 April 2018).
- Smart Procurement. 2011. SA public procurement: poor value for money. Available from http://www.smartprocurement.co.za/archives/sa_public_procurement_-_poor_value_for_money.php: (Accessed 25 April 2018).

- Smith, P.R and Zook, Z. 2011. *Marketing Communications: Integrating offline and online with social media*. 5th ed. London: Kognan Page.
- South Africa. 1999. Public Finance Management Act 1999 (online). Available: http://PFMA.gov.za (Accessed 12 November 2015).
- South African Post Office (2018). Home (online). Available from www.sapo.gov.za (Accessed 15 June 2018).
- South African Social Security Agency (2018). Summary Report (Online). Available from: http://www.sassa/governance/ar.asp (Accessed 25 January 2018).
- Spicer, M. 2016. 'The Business Government Relationship: What Has Gone Wrong?' *The Journal of the Helen Suzman Foundation*, online. Available at: https://hsf.org.za/publications/focus/focus-78-the-economy-1/michael-spicer.pdf (Accessed 01 June 2020).
- Statistic South Africa (2013). HOME (online), Available: http://www.statssa.gov.za/ (Accessed 3 May 2018).
- Taponen, S. 2017. Improving the efficiency of public sector service delivery through outsourcing and management. PhD, Aalto University.
- Tidwell, W.2012. Writing a conceptual framework. (Online). Available from: http://www.slideshare.net/wtidwell/writing-a-conceptual-framework (Accessed 1 August 2020).
- Trent University. 2014. Theoretical frameworks. Available from: http://www.trentu.ca/history/workbook/theoreticalframeworks.php (Accessed 15 April 2018).
- Tucker, M. L., Powell, K., and Meyer, G. 1995. Qualitative research in business communication: A review and analysis. *Journal of Business Communication*, 32, 383.
- Tukuta, M. and Saruchera, F. (2015). Challenges facing procurement professionals in developing economies: *Unlocking value through professional international purchasing*. 9(1): 123-130.
- Turley, L. and Perera, O. 2014. "Implementing sustainable public procurement in South Africa: Where to start?", The International Institute for Sustainable Development.
- Kreuser, C., and C. Newman (2018). 'Total Factor Productivity in South African Manufacturing Firms'. South African Journal of Economics, 86: 40–78
- Kashap, S. 2004. Public procurement as a social, economic and political policy, International Public Procurement Conference Proceedings, 3(1): 133-147.
- United Nations Development Plan. 2007. Procurement aspects of introducing ICTs solutions in electrical processes. (Online). Available from: http://www.za.undp.org/content/south_africa/en/home/procurement.html. (Accessed 25 April 2018).
- United Nations Development Plan. 2016. (Online)
 Available from: https://www.undp.org/content/undp/en/home.html (Accessed 25 April 2018).
- Uyarra, E. and Flanagan, K. 2009. Understanding the innovation impacts of public procurement. *Manchester Business School*, 574.

- Uzma, N. and Samreen, L. 2016. Impact of Customer Self Concept and Lifestyle on Luxury Goods Purchases: A Case of Females of Karachi. *Arab Journal Business Management*, 6: 192.
- Van Weele, A.J. (2009). *Purchasing and supply chain management: analysis, strategy, planning and practice*, 6th ed. London: Thomson Learning.
- Van Zyl, D.C.2006.Strategic supply chain management by Matatiele Municipality. Master's Degree, University of Stellenbosch.
- Vintar, M and Stanimirovic, Evaluating of impact of outsourcing on efficiency of public sector organisations (Online). Available from: https://www.dergipark.org.tr/ijbms/issue/26069/274737 (Accessed 04 April 2019).
- Walker, H and Brammer, S. 2009. Sustainable procurement in the United Kingdom public sector. *An International Journal*, 14(2): 128-137.
- Watermeyer, R.B. 2011. Regulating public procurement in Southern Africa through international and national standards: *Public procurement regulation in Africa conference*. South Africa, 1-11. March Stellenbosch. South Africa.
- Wernerfelt, B. 1984. A resource-based view of the firm. Strategic Management Journal, 5(2), 171-180.
- Wiesul, K. 2015. *How to outsource without going broke or insane* (online). Available from: http://www.forbes.com/sites/steveden ni ng/20 13/01 /21 what -went-wrong-at-Boeing (Accessed 07 May 2018).
- Wiid, J and C. Diggines, 2010. Marketing research. South Africa. Juta and Company Ltd.
- Williamson, O.E. 1979. Transaction-Cost Economics: The Governance of Contractual Relations. *The Journal of Law and Economics*, 22(2): 233-261.
- Yin, R.K. 1981. The Case-Study of a serious research strategy. The Case Study Institute, 3(1):97-114.
- Zack, M.H. 2000. Competing on knowledge. 2nd ed. New York: Macmillan.
- Zsidisin, G.A. 2003. "Managerial perceptions of supply risk", *Journal of Supply Chain Management*, 39(1): 14-25.

Appendices

Appendix A: INTERVIEW GUIDE QUESTIONS FOR INTERVIEWS AND FOCUS GROUPS

I am a master's student registered with the University of KwaZulu-Natal in South Africa at the Pietermaritzburg campus. I am currently conducting a study titled: *an examination of the barriers in the effective use of outsourcing in the public sector*. I will be grateful if you would take time to answer the following questions. It will only take 20-25 minutes to answer. The data you provide will be treated in the strictest confidence and will only be utilised for academic research purposes. Your anonymity is guaranteed and no reference to any respondent will be made within the research report. I look forward to your favourable consideration with regard to participating in the completing questionnaire at your earliest convenience.

- 1. Do you think the number of staff working in the supply chain management is sufficient given the size and needs of the organisation?
- 2. Is there a clearly defined strategy in the processes and procedures related to outsourcing and how do you rate them? Please explain.
- 3. Are the department's outsourcing of functions ensuring organisational continuity, if not why does the department outsource?
- 4. Does management have proper qualifications and training in relation to outsourcing?
- 5. Does the organisation readily adopt new technologies to improve on the outsourcing processes?
- 6. How are managers and staff rewarded for achieving organisation's objectives and improving the value delivered to clients?
- 7. How does a department ensure that outsourced programs and services are unique and more highly valued than alternatives?
- 8. How would you rate the impact of outsourcing to the organisational goals and objectives?
- 9. Does the organisation effectively communicate the value of outsourcing to all the stakeholders?
- 10. How does the organisation measure results? Do they actively identify areas for improvement?
- 11. How does the department ensure that client satisfaction is maintained at a high level?
- 12. How do you rate the impact of the control measures on attitudes and behaviors of the employees?

Appendix B: Informed Consent form

UNIVERSITY OF KWAZULU-NATAL

School of Management, IT and Governance

Dear Respondents

Research dissertation: Master in Commerce: Management

Researcher: Mondli Msiya (0813255168) email:m.msiya@yahoo.com

Supervisor: Dr O. E. Okeke-Uzodike (031 260 5736) email: okekeuzodikeo@ukzn.ac.za

Research Office: Humanities & Social Sciences Research Ethics Administration, Govan Mbeki

Building, Westville Campus, Tel: + 27 (0)31 260 8350, Email: hssreclms@ukzn.ac.za

I, Mondli Msiya, am currently registered at the University of KwaZulu-Natal PMB campus for studies

leading to a MCOM in Management. I am carrying out a study at the South African Social Security

Agency at Umzumbe Local Offices. The study is titled: An examination of the barriers in the effective

use of outsourcing in the public sector.

The purpose of this study is to examine the cost-effectiveness of outsourcing marketing functions in the

public sector. To achieve the objectives of this research, the researcher needs assistance from you to

answer the questions. The attached interview questions will take about 15 minutes to complete. Your

participation in this project is voluntary. Please rest assured that your responses will be treated with the

utmost confidentiality and will not be divulged to any other party. The researcher will ensure that the

information that you will provide will not be used for any other purpose except for their research

investigation only. Anonymity will be ensured in this research. Data from the respondents cannot be

linked to a specific individual. The researcher will never injure or harm the people being studied,

regardless of whether they volunteered to be part of the study or not. Privacy will be ensured in this

study. Your participation will be highly appreciated.

If you have any questions or concerns about participating in this study, please contact me or my

supervisor.

Sincerely

Researcher's signature Date

This page is to be retained by participant

75

UNIVERSITY OF KWAZULU-NATAL

School of Management, IT and Governance

Researcher: Mondli Msiya (Telephone: 0813255168), (Email: <u>215080154@stu.ukzn.ac.za</u>) Supervisor: Dr O.E. Okeke-Uzodike, (Telephone number: 031 2605736), (Email: <u>okekeuzodike@ukzn.ac.za</u>) Research Office: Humanities & Social Sciences Research Ethics Administration, Govan Mbeki Building, Westville Campus, Tel: + 27 (0)31 260 8350,

Email: hssreclms@ukzn.ac.za

Protocol reference number: HSS/0014/018M

Informed consent

Signature of participant Date



25 August 2020

Mr Mondli Victor Msiya (215080154) School of Management, IT & Governance **Pietermaritzburg Campus**

Dear Mr Msiya,

Protocol reference number: HSS/0014/018M

New Project title: An examination of the barriers in the effective use of outsourcing in the public sector

Approval Notification – Amendment Application

This letter serves to notify you that your application and request for an amendment received on 24 August 2020 has now been approved as follows:

Change in title

Any alterations to the approved research protocol i.e. Questionnaire/Interview Schedule, Informed Consent Form; Title of the Project, Location of the Study must be reviewed and approved through an amendment /modification prior to its implementation. In case you have further queries, please quote the above reference number.

PLEASE NOTE: Research data should be securely stored in the discipline/department for a period of 5 years.

All research conducted during the COVID-19 period must adhere to the national and UKZN guidelines.

Best wishes for the successful completion of your research protocol.

Yours faithfully



Professor Dipane Hlalele (Chair)

/dd

Cc Supervisor: Dr OE Okeke-Uzodike

Cc Academic Leader Research: Professor Isabel Martins

Cc School Administrator: Ms Jessica Chetty

Humanities & Social Sciences Research Ethics Committee UKZN Research Ethics Office Westville Campus, Govan Mbeki Building Postal Address: Private Bag X54001, Durban 4000 Tel: +27 31 260 8350 / 4557 / 3587 Website: http://research.ukzn.ac.za/Research-Ethics/ ood Howard College Medical School

Founding Campuses:

Edgewood

Pietermaritzburg

Westville

Dissertation

ORIGINALITY REPORT

3% 3% % % % SIMILARITY INDEX INTERNET SOURCES PUBLICATIONS STUDENT PAPERS

PRIMARY SOURCES

uir.unisa.ac.za
Internet Source
2%

ukznextendedlearning.com
Internet Source
1 %

Exclude quotes On Exclude matches < 100 words

Exclude bibliography On